DOCUMENT RESUME

ED 223 737	UD 022 497
TITLE	Resource Allocation Study. Final Report. Draft.
INSTITUTION	Alum Rock Union Elementary School District, San Jose, Calif.
SPONS AGENCY	California State Dept. of Education, Sacramento.;
· · ·	National Inst. of Education (ED), Washington, DC.
PUB DATE	82
CONTRACT	400-75-0060
NOTE	122p.; Best copy available.
PUB TYPE	Reports - Research/Technical (143)
EDRS PRICE	MF01/PC05 Plus Postage.
DESCRIPTORS	*Class Activities; *Classroom Environment; Class Size; Compensatory Education; Disadvantaged Youth; Elementary Education; Federal Programs; *Reading Achievement: *Resource Allocation; Student Behavior;

IDENTIFIERS

*Teaching Methods *Alum Rock Union School District CA; Elementary Secondary Education Act Title I

Student Teacher Relationship; Teacher Behavior;

ABSTRACT

A study in 18 schools in the Alum Rock (California) Union Elementary School District investigated the educational effects of differential distribution of compensatory resources and services to educationally disadvantaged pupils. The study focused on the relative effects on reading skills of two methods for resource targeting: (1) concentration, or the targeting of additional resources and services only to those pupils who meet the entitlement criteria in a school; and (2) saturation, or the distribution of additional resources and services to any and all pupils in the school. In addition, the study investigated the contextual and instructional conditions that might account for differences in reading skills. Classroom observations were conducted to examine (1) types of teacher and student activities; (2) time spent on instruction; (3) teacher-pupil interaction; (4) materials; (5) teachers' interpersonal styles; (6) instructional modes; and (7) group size. Data analysis indicated that concentrated services produced higher mean reading scores than saturated services among fourth graders, although no significant differences were found among second-graders; and that more time spent in noninstructional activities tended to lower reading scores. In general, however, the analysis did not identify the instructional components associated with the benefits of concentrated compensatory services. (Author/MJL)

DRAFT

U.S. DEPARTMENT OF EDUCATION NATIONAL INSTITUTE OF EDUCATION EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC) document has been reproduced as received from the person or organization

Minor changes have been made to improve

Points of view or opinions stated in this document do not necessarily represent official NIE

friginating it.

reproduction quality.

position or policy,

BEST COPY AVAILABLE

E022373

49-220 QU

ERIC

FINAL REPORT **RESOURCE** ALLOCATION STUDY

SUBMITTED TO THE NATIONAL INSTITUTE OF EDUCATION

Prepared by: Alum Rock Union School Distric 2930 Gay Avenue San Jose, California 95127

William J. Jefferds Superintendent

California State Dept. of Education

400-75-0060

TABLE OF CONTENTS

		Page(s)
I. IN	RODUCTION	1 - 7
· A.	General	
Β.	Overview of Observational Component	
_ C.	Overview of Analytical Component	
4		
	C ODCEDUATION CONDONENT	8_0/
	E OBSERVATION COMPONENT	0-9
Α.	Ubserver framing kernability	
III. CL	ASSROOM OBSERVATIONS	10 - 27
Α.	Instrumentation and Data Collection	
Β.	Classroom Demographics	
;	 Class Size and Pupil-Adult Ratio Pupil Teacher Grouping Size 	
•	 Use of Instructional Resources Outside the Classroom Classroom Composition 	
с.		
	1. Teacher Roles 2. Pupil Roles	
• •	3. Teacher and Pupil Time Engaged in Instructional Activities	
·	4. Engaged Time 5. Instructional Style	
	6. Instructional Materials 7. Teachers' Interpersonal Style	
	7. Teachers Interpersonal objie	
IV. IN	DIVIDUAL PUPIL OBSERVATIONS	28 - 36
Α.	Sample	
В.		
	 Engaged Time Instructional Modes 	
	3. Instructional Materials	
	4. Group Size	
V. AN	ALYSIS OF EFFECTS	37 - 44
. A.		
- B.	Analyses of the Effects at the Teacher/Class Level	
	 Results of Covariance on the 28 Second Grade Classes Results of Analysis of Covariance on the 28 Grade 4 Classes 	· .



-	f	Page(s)
VI.:"	RESULTS FROM ANALYSES OF THE WITHIN-CLASS PUPIL SAMPLES	45 - 56
	A. Grade 2 Regression Results	
	B. Grade 4 Regression Results	
!	C. Regressions Without EDY Status as a Context Variable	
	D. Analysis of Covariance Results 1. Grade 2 Spring 1978 Outcomes 2. Grade 2 Difference Score Analyses 3. Results of the Grade 4 Analyses	· ·
-		
II.	MULTIPLE REGRESSION ANALYSES RESULTS	
	 A. Grade 4 Class-Level Regressions 1. Total Reading Achievement 2. Word Knowledge 3. Reading Comprehension 	
	 B. Grade 2 Classroom-Level Regressions 1. Word Knowledge 2. Reading Comprehension 3. Word Analysis 4. Total Reading Score 	4
11.	SUMMARY CONCLUSIONS	67 - 7
IX.	APPENDICES	74 - 10
	APPENDIX A - DEFINITION OF ROLE TYPES	
	APPENDIX A - DEFINITION OF ROLL TIPES APPENDIX B - ANALYSIS VARIABLES	
	APPENDIX C - INTERCORRELATION MATRIX FOR CLASSROOM-LEVEL ANALYSIS VARIABLES	
	APPENDIX D - SUMMARY OF REGRESSION ANALYSIS	
	APPENDIX E - REGRESSION RESULTS WITHOUT EDY STATUS	
	APPENDIX F - FOUR-WAY ANALYSIS OF COVARIANCE FOR SECOND GRADE PUPILS	
	APPENDIX G - DIFFERENCE SCORE ANALYSIS FOR SECOND GRADE PUPILS	, ,
	APPENDIX H - FOUR-WAY ANALYSIS OF COVARIANCE FOR FOURTH GRADE PUPILS	
-	APPENDIX I - DIFFERENCE SCORE ANALYSIS	

á

ERIC Printex Provided by ERIC

INTRODUCTION

GENERAL

This report presents the findings of a study of the relative educational effects of differential distribution of compensatory resources and services to educationally disadvantaged pupils in the Alum Rock Union Elementary School District. The National Institute of Education (NIE) sanctioned and supported this study as part of a study of improvements in ESEA Title I. Additional authorization came from State and local agencies. Alum Rock Union Elementary School District participated as one of 11 national demonstration sites with specific interest in assessing the relative educational effects of variations in school wide targeting of compensatory services. The effects of the concentration and saturation methods provides the general focus of this report.

Two methods for resource targeting were designed to satisfy the study objective:

- (a) <u>Concentration</u>-or the targeting of additional resources and services
 - only to those pupils who meet the entitlement criteria in a school.
- (b) <u>Saturation</u>--or the distribution of additional resources and services to any and all pupils in a school.

The data base used in assessing the relative effects of these two resource provision conditions was provided by establishing a matched sample of 18 schools which were randomly assigned to "saturation" or "concentration" of EDY resources. In the Fall of 1976 (school year 1976-77), administration and faculty at each school were provided guidelines for implementation of their respective resource treatment: concentration or saturation. To avoid administrative and legal problems associated with compensatory programs, necessary waivers from State and Federal agencies were solicited and obtained prior to actual implementation of the treatment. In fact, resources from these waived programs (ECE, SB90 and Title I) were pooled at the district level and reallocated to participating buildings on a direct EDY targeting basis. Therefore, a uniform EDY resource share per pupil was established, and the number of EDY pupils in a given building determined the resource allocation (i.e., number of shares) for that building. It should be noted that a pepil was defined as EDY based on reading performance on a standardized achievement test: viz. at or below the 50th percentile on the Metro 70 (Harcourt, Brace, Jovanovich, 1972).

1.

 $\overline{\mathbf{5}}$

Many issues contemplated by this study are extremely complex. To address those issues, a multifaceted research model was designed consisting of two components: <u>The Observational Component</u>, and the <u>Analysis of Effect</u>.⁴ The following questions provided guidance in the design and implementation of the research model:

General Questions:

-

- Does saturation or concentration of compensatory resources and services relate to meaningful and reliable differences in basic reading skill acquisitions for either EDY pupils, non-EDY pupils, or both?
- 2. From a logistical standpoint, what is the utility of saturation vs. concentration as a means for improvement in the acquisition of reading skills?
- 3. To what extent do indicators of concentration vs. saturation relate to <u>improvements</u> in the acquisition of reading skills by EDY pupils?
- 4. To what extent does saturation vs. concentration produce general educational benefits, in terms of overall (class level) achievement in reading skills?
- 5. What are the 'implications of findings from this study for:
 - 1) Inservicing policy at the District level?
 - 2) Guidelines for targeting and coordination/integration of overlapping programs at the State level?
 - 3) Resource allocation, targeting and compensatory service policy guidelines at the Federal level?
- 6. What are the implications of unanticipated findings (or lack of anticipated findings) for designing follow-up studies, including reanalyses of the current data?

2.

ΰ

Observational Questions

- 1. What are the resources and processes used for reading instruction?
- 2. How are these processes and resources used differently in saturated vs. concentrated classes?
- 3. Do teachers use different materials and/or methods with EDY as opposed to non-EDY pupils? In the concentrated class? In the saturated class?
- 4. What are the contextual and procedural (instructional) conditions which account for differences in reading skill attainment?
- 5. How were the guidelines for concentration or saturation of compensatory services and resources actually implemented in the demonstration schools as manifest by instructional and administrative staff attitudes and behaviors?

- Analysis Questions

- What are the more salient characteristics (in a statistical sense) of teacher effectiveness, in terms developed by this study, and how do these effectiveness characteristics compare with those identified through other studies?
- 2. What are the important resources and service components and their use configuration which best accounts for improved reading skill attainment among EDY pupils?
- 3. How do instructional effects or saturation vs. concentration differ at individual pupil vs. class aggregated levels? That is, to what extent must class (or school) context variable be taken into account in understanding the impacts of the implementation of resource and service?

This report presents discussions pertaining to the two components of the research mode. In addition, a brief summation of the overall findings is presented.

OVERVIEW OF OBSERVATIONAL COMPONENT

The observational component of the research model was designed to accomplish three general objectives. First, a description of the resources and processes used during reading instruction was desired. Secondly, difference in how the two treatments (concentration vs. saturation) were implemented was to be observed in terms of what resources and processes were used. Finally, instructional process variables were generated and defined to become part of the basis for analysis in the analytical component.

To accomplish these ends, observation model was developed consisting of two observation instruments: <u>Classroom Observation Treatment</u> and <u>Individual Student Instrument</u>.

The CLASSROOM OBSERVATION INSTRUMENT was designed to obtain information about activities and materials used by the teacher and the class as a unit. Most of the observations focused on both the teacher's interaction with the students and how the teacher utilized the materials in the classroom. The instrument consisted of two ten-minute teacher-focused observations episodes, separated by a ten-minute observation focusing on the class in general. Most of the information for study purposes was obtained during the teacher-focused observation.

During the classroom observation, the observer examined the relationships between EDY funding strategies and patterns of classroom processes such as teacher decision making; teacher role orientation, classroom content, classroom organization, and patterns of student interaction with staff, other students, and material resources. This instrument examined classroom implementation patterns in terms of the relationship between funding strategy (treatment) and student outcomes. The classroom observation instrument provided a depiction of differences in student classroom experiences under the two funding conditions, and a determination of the relationship between the differences in classroom processes and student outcomes under the two funding conditions. This instrument did not contemplate generalizations about funds in any given school or classroom since the analysis was designed for district wide results.

*Measures of student outcomes were gathered from student scores on achievement tests (n=2,100), teacher assessment of the proportion of students' objectives accomplished for a small portion of the class, and/observation of task engagement.

The INDIVIDUAL STUDENT INSTRUMENT was designed to obtain information about how specific students were involved in class activities and what materials these students used. The observer focused on individual students who were pre-selected according to grade level, sex, ethnicity, EDY status and the type of class treatment in which the students were involved. Each of four students per classroom was observed for approximately 30 minutes.

The individual student observations were designed to obtain information describing instructional resources and processes used by teachers in the two treatment conditions. Unlike the teacher-focused observations, the individual student observations provided unique information about the implementation of the saturated or concentrated treatment, specifically, in determining how teachers differentiate resources and processes used on the basis of pupil's EDY status.

During the individual student observation, the observer examined the relationships between EDY funding strategies and patterns of student interactions within the class. The observer noted student role orientation, how student was involved in classroom grouping, how student used funded resources and additional relationships and patterns concerning student use of EDY materials and resources.

The combined information obtained from the use of these two observation instruments provided for the generation of instructional process variables. These variables were part of the data base for the analytical component of the research model.

OVERVIEW OF ANALYTICAL COMPONENT

Two basic issues guided the analysis of study data for evidence of treatment effects:

- Does saturation or concentration of compensatory resources and services-to the extent such occurred in this study--relate to reliable and meaningful differences in basic reading skill attainment?
- 2) What are the contextual and procedural (instructional) conditions which account for differences in reading skill attainment. $\frac{1}{2}$

The first question focuses on the effectiveness of the implementation of the two treatments. More succinctly, which treatment of allocating educational resources and services provides the highest pupil reading achievement?

An additional objective was to determine the central and peripheral effects of the allocations of these treatments on class practices and pupil learning (reading, as measured by MAT). The second analysis question focuses on the more general domain of instructional effects. The objective here was to determine what the characteristics of pupils, resources, and instructional procedures which taken to-gether, accounted for learning outcomes (i.e., reading skills, as measured by the MAT).

To properly address these two issues and the other general study issues, the analysis was divided into four parts. Part I contains a discussion about the development of analysis variables. Part II contains an examination of the data for evidence of effects due to saturation or concentration, using data collected at the class-level. Results of these analyses are reported separately for the second and fourth grades. Part III contains an extension of the analysis to include information on the relative effects of the two alternative modes of delivering compensatory resources and services at the individual pupil level. This provides an examination of pre-post test patterns in terms of degree of educational disadvantage, ethnicity, gender, and the interactions of these conditions with the alternative "treatments" as implemented by the teachers. Finally, Part IV deals with the more general question of how this information regarding contextual and instructional processes used in the class explain outcomes observed at the class level.

In Part I, statistical tests of probability were used in the effects of saturation vs. concentration; however, greater emphasis was placed on identifying and better understanding the proximal (near) and distal (far) consequences of this attempted intervention. Additional emphasis was placed on evaluating such effects against alternative input-process-outcome patterns detected in the data. Variables designed to accomplish these analyses were derived from consideration of the fundamental issues regarding saturation vs. concentration of compensatory resources and the results of contemporary teacher effectiveness research.

In Part II, the information obtained either during the classroom observations, through interviews and test data, or through additional methods was scrutinized to determine whether complete data on all relevant measures were available for each class. The requisite complete data set was found for 56 of the teacher/class units observed.

6.LU

ERIC Full Taxt Provided by EFIC Two types of conditional analysis were subsequently performed on these data. Both types are based on the general linear hypothesis. First, two-way analyses of covariance were performed on each of the four outcome variables (total reading, word knowledge subscore, word analysis subscore, and reading subscore) within grade level (second and fourth). The treatment condition (saturated or concentrated) was used as the between-group variable. The five context or process measures which showed the closest relationship were treated as covariables. Secondly, multiple linear regressions were performed on these data. The mean reading achievement scores were regressed on several combinations of context and process variables to identify the most significant determinants of outcome score variance. Through this analytical technique, an assessment of the instructional effects of several process variables was possible when the effects of context variables were sufficiently controlled.

In Part III, multiple linear regressions were performed on samples within gradelevel to identify reliable context and process covariates of pupil achievement. The available measures for each pupil included prescore (previous MAT standard scores for sub and total tests), design variables (EDY status, resource treatment, ethnicity, gender, and a set of process observation descriptors. Consequently, within each grade level post-scores were regressed on available process and context variables, including the corresponding prescore. The identification of relevant process covariates of outcomes was enhanced by attaching differential weight factors to the process variables in the stepwise procedures.

In Part IV, additional analysis was performed to attempt to evaluate the instructional components and other features associated with the possible benefits (in terms of reading achievement) of concentrating compensatory services and resources. To facilitate the analyses and evaluation of these features, a series of stepwise multiple linear regressions were performed on outcome measu-es within each grade-level sample. Basically, this analysis attempts to discover what are the process and context characteristics at the classroom level which best account for differences in mean achievement?

11

THE OBSERVATION COMPONENT

The observational phase of this study was designed to achieve three

objectives:

-

- To describe and define resources and processes used for reading instruction;
- To detect differences in how the resources and processes are used between the saturated and concentrated classes and to determine how these differences are related to the implementation of the two treatments;
- To generate instructional process variables which, when integrated with interview and test data, provide a basis for studying the relationships between the processes used and the pupil outcomes. (See Analysis of Effects section for discussion of process/outcome study.)

The findings relating to these objectives are presented in this section. Additional analyses utilizing these findings are discussed in subsequent chapters.

All observations were conducted during reading instruction. The rationale for this is twofold. The primary reason is basically methodological. Since one of the objectives was to determine differences both within and between class treatment types (e.g., differences between individual children, variations over time), it was necessary to minimize the inherently convoluted effects on the data which would have resulted had the observations also been taken during math, science or art instruction. The second reason is primarily political. The relative success or failure of providing basic reading skills to elementary students is presently a topic of widespread attention and concern. Consequently, improving the effectiveness of reading instruction remains a high priority for Title I and other compensatory education programs.

Two observation instruments were developed for this study. The CLASSROOM OBSERVATION INSTRUMENT was designed to obtain information about the activities and materials used by the teacher and the class as a unit. Most of the observations focused on both the teacher's interaction with the students and how the teacher used the materials in the classroom. The INDIVIDUAL STUDENT INSTRUMENT was designed to obtain information about how specific students were involved in these or other

activities and what materials those students used. This combined observation procedure provided a multiple perspective on the phenomenon of reading instruction as it occurred within the second and fourth grade classrooms.

Observer Training and Reliability

Six substitute teachers participated as observers for this study along with two supervisory observers. The six were selected based on these criteria: analytical skills, superior memory, prior classroom observation experience, the ability to objectively stay within the study's definitional bounds, and a willingness to work unusual part-time hours.

Each observer received a minimum 100 hours of training before taking observation in the classroom. The training involved lecture discussions, homework and review of each section of the instruments. Role playing and classroom video tape analysis supplemented these activities.

In addition, the observers conducted practice observations in over 30 different classrooms. During these practice sessions, the reliability of the observer was evaluated. The observer had to attain at least 90% proficiency on the reliability test in order to continue in the study. The reliability of observer was evaluated throughout the study.

Unfortunately, insufficient time had been allocated to test the six observers for reliability, and it was necessary to extend this testing process into the first week of the actual observations. Consequently, to assure reliability of the data collected during this period, trainees were required to conduct classroom observations under the supervision of a reliable observer. Observers were allowed to conduct observations alone only after they had demonstrated satisfactory reliability. Every trainee had repeatedly and satisfactorily demonstrated reliability by the end of the first week of actual observations.

Classroom Observations

Instrumentation and Data Collection

The Classroom Observation instrument was designed to include two ten-minute observation episodes during which the observer recorded detailed information about activities and materials used by the teacher and students working with the teacher. The pupils working either with an aide or on their own (e.g., self-instruction activities not supervised by teacher) were not observed during the teacher-focused observation. The observer recorded information including: the number of/pupils in the teacher's group, the nature and duration of the teacher activities, the nature and duration of the pupil activities, the type of materials used, and the frequency of approval and disapproval of pupil work or behavior.

The two ten-minute observations were separated by a ten-minute class observation during which the observer recorded less detailed information about activities and materials used in the class and pupil grouping patterns throughout the classroom. This information provided background and supplementary data and was not generally included ir the analysis presented in this report. Therefore, the information obtained during the teacher-focused observation provides the bulk of the data presented for the classrooc observation component.

Only 56 second and fourth grade classes from the 72 originally considered provide the requisite complete data set (i.e., teacher and principal interviews, classroom and individual observations, and test results for the teacher's previous and present class). Each of these classes was observed for two ten-minute episodes during the reading instruction period on four separate days; therefore, providing a total of eight separate ten-minute classroom observation episodes. The data from these eight observation episodes were combined to calculate measure of central tendency (mean) and variation (standard deviation) for the class.

The development of analysis variable based on the data from the Classroom Observation Instrument is presented in this chapter. The results of descriptive analysi used to examine the types of resources and processes used for reading instruction is als presented. In addition, the result of analysis of variance, conducted to detect differences in resources and processes across treatment groups and grade levels, is reported.

Classroom Demographics

Class Size and Pupil-Adult Ratio

The average number of pupils present during the observation period was 17.8. The size of a class ranged from 5 to 32 pupils; however, approximately half the classes consisted of 12-24 pupils. The average number of pupils present during reading instruction was approximately the same for saturated and concentrated classes.

One reason for the wide range in class size during reading instruction was the use of different scheduling procedures. Approximately one-third of the classes operated unde: a regular total class schedule. The staggered schedule was adopted in the remaining twothirds of the classes. The classes using this procedure divided the students into two groups based on reading aptitude or some other criterion. One group arrived at school on hour early for reading instruction. This group left an hour earlier than the second group, which had reading instruction at the end of the day. The staggered schedule was used equally by concentrated and saturated classes. This type of schedule significantly reduced class size during reading instruction; however, it lengthened the teaching day. The classes using the staggered schedule succeeded in lowering the pupil-adult ratio during reading instruction.

An additional method used to lower the pupil-adult ratio during reading instruction was the use of instructional aides. Aides were present during 44% of the observations. Aides were found more often in saturated classes (53%) than in concentrated classes (36%) and considerably more often in total-class situations (73%) than in classes using the staggered schedule (28%). (Both the treatment group and schedule differences are significant at $p \le .05$, the significance level used in this study as the criterion for identifying reliable differences.) Therefore, aides were most often present in saturated total-class situations--classrooms in which larger numbers of pupils were present and aides were permitted to work with any pupil. Conversely, aides were, least likely to be found in classes in which the staggered schedule was used to reduce class size and the aide was restricted to working with pupils classified as EDY.

Average size of the teacher's total class was about 28 for both the concentrated and saturated groups.

 \mathbf{n} .

The use of the staggered schedule and the use of aides enable the district to realize an average pupil-adult ratio of 13:1 for reading instruction in the classroom observed. However, because of the widespread use of aides in classes operating under the total-class schedule, pupil-adult ratio did not differ as widely between the staggered classes (average of 12:1) and total classes (average of 165) as might have been expected. Reliable differences across treatment groups were not found.

Pupil Teacher Grouping Size

Teachers worked with an average of about 10 pupils during a ten-minute episode. This represents slightly over half the pupils present. Group size did not differ significantly across treatment types or grade levels.

Use of Instructional Resources Outside the Classroom

Under both treatment conditions, instructional resources were rarely used outside the classroom during reading instruction. One or more pupils were sent to a reading specialist or a resource center in only 7% of the observations. This finding, however, does not indicate that these resources have not been fully utilized, primarily because federal regulations specify that resources purchased with Title I funds should be used to supplement rather than supplant basic instruction. Therefore, limited use of thse resources during the basic reading instruction period is in accord with these regulations.

Classroom Composition

Saturated and concentrated classes in the sample were similar in composition. In both groups, approximately two-thirds of the pupils present during the observations were classified as EDY. The average ethnic composition in both groups was 58% Spanish surname, 23% caucasian, 12% black and 7% other. The average age was 8 years 11 months for second-grade and 10 years 4 months for fourth grade. In addition, both groups consisted of equal proportions of boys and girls.



Descriptive Results -- Teacher-focused Observations

Teacher Roles

The observer recorded the teacher's activities during the ten-minute teacherfocused observation using a role code. (See Appendix A.1 for operational definitions.) The analysis variables developed from the teacher role code include measure of role diversity (average number of different roles per episode), and role types (instructional vs. nominstructional, directive vs. self-instruction, interactive vs. facilitative).

The observer could record up to six roles per observation episode; however, the average number observed was moderately low (2.26).^{*} In only 10% of the teacher-focused observations were more than three roles observed. The maximum number observed was five. Diversity of teacher roles did not differ reliably across treatment group or grade levels.

Virtually all of the 25 roles codified were observed in at least one observation episode; however, a relatively few roles accounted for the substantial percentage of all roles observed.⁺ The most common teacher activities observed were oral or silent reading and reviewing (24%),^{**} drill (23%), classroom management (15%), and assigning tasks (9%). Together these four activities represented 71% of all teacher roles observed. These roles were predominant across treatment groups and grade levels.

Roles could be interspersed, as in the case of a teacher who alternated between making assignments and drill activities. However, a given role code was recorded only once during an episode. The time recorded for the role reflected total number of minutes across all occurrences of the role within the episode.

Frequency of occurrence for each of the 25 roles is shown in Appendix B.3.

This group activity is actually a combination of three codes: oral/silent reading, reviewing, and oral/silent reading with review. The distinction between the first two codes and the third is essentially one of pacing. Observers used the third code when the teacher continually alternated between asking pupils to read passages aloud or silently and asking questions about the passages. When either the reading or the questioning persisted for two minutes or longer, without interruption by the other, the component activity was coded.

Pupil Roles

The roles in which the pupils were involved while working with the teachers were also recorded. (See Appendix A.2 for operational definition.) The observer could observe up to six different roles per episode, although the average number observed per episode was moderately low (2.59). In only 20% of the groups observed were more than three pupil roles recorded. The diversity of pupil roles did not differ reliably across treatment groups or grade levels.

As with the teacher roles, numerous pupil roles were observed with only a few predominating.⁺ The pupils' predominating roles and the teachers' predominating roles were related to involvement by both pupil and teacher in the same activities; therefore, many of the pupil roles were counterpart to the teacher roles. The most common activitie: involving pupils working with the teacher were oral/silent reading and review (22%) and drill (20%). Receiving assignments and participation in classroom, management activities represented 8% and 9% of all pupil roles, respectively. Two additional roles were fairly common among pupils; seatwork (completing assignments--11%) and transition (waiting for a new task or the teacher's attention--6%). Together these activities accounted for three-fourths of all the pupil roles observed.

Teacher and Pupil Time Engaged in Instructional Activities

In the previous two sections, the frequency of specific teacher and pupil roles was examined without regard to the amount of time spent in those roles. The observers did, however, record the number of minutes associated with each role. Two dimensions

Multiple pupil roles were recorded when the whole teacher's group switched from one activity to another or when different members of the group were simultaneously engaged in different roles.

Relative frequencies of all pupil roles are shown in Appendix B.4.

14.

were developed both to examine any patterns associated with how the teachers and the pupils used their time during reading instruction and to determine whether any relationship existed between those patterns and test outcomes. The first dimension, engaged time was a measure of the proportion of time devoted to instruction and activities directly related to instruction. The second dimension, instructional style, was a characterizatic of the activity level or mode of instruction associated with each role observed.

Engaged Time

In order to measure the proportion of time devoted to instruction and the association with the instructor, all of the teacher and pupil roles were classified as instructional or noninstructional in nature.^{*} Then the proportion of teacher and pupil time associated with instructional roles was calculated for each observation episode. Complete data from all observations of a given class were combined to derive measures of average pupil and teacher engaged time. This measure was found for all 56 classes.

The results presented in Table <u>1</u> indicate consistently high proportions of engaged time. Generally, pupils and teachers spent about 90% of their time in activitie of an instructional nature. This indicates that out of the 80 minutes of total observation time per classroom, an average of less than nine was devoted to classroom managemen discipline, and other activities not directly related to instruction. This pattern of high engaged time is consistent across treatment groups and grade levels.

Table 1

	A	verage Perc	ent of Tim	e	
	Satu	rated	Concen	trated	A11
	Grade 2	Grade 4	Grade 2	Grade 4	<u>Classes</u>
Engaged TimeTeachers	90.97%	87.31%	91.29%	83.42%	88 - 29%
Engaged TimePupils	89.60	87.99	90.24	87.97 -	88.94
(Number of Cases	(15)	(14)	(13)	(14)	(56)

Teacher and Pupil Engaged Time during Teacher-Focused Observations

A breakdown of instructional vs. noninstructional roles is shown in Appendices A.1 and A.2.

ERI

Instructional Style

Activities levels or modes of instruction were developed to facilitate the examination of patterns of time utilization associated with each role observed. Teacher roles were grouped into five nonoverlapping modes, each of which is described below. The specific roles included in each category, and their relative prominence within the category, are shown in Table 2.

<u>Teacher directive roles</u>. In these roles, the teacher is the primary actor. Where pupils are involved, they are typically receiving directions or other information from the teacher, with no immediate response or participation (other than listening) required on their part. Teachers observed in directive roles were most often performing classroom management tasks, making assignments, or instructing (lecturing). Other activities included in this category but observed much less often are reading aloud, discipline, and praise.

<u>Teacher-initiated interactive roles</u>. In these roles, the teacher also acts as instructional leader, but pupils are assumed to take a more active part. Basically these roles represent several variations on a question-and-response format, with pupils making frequent group or individual responses in these interactive roles. Teachers were usually leading drill (for example, on wordattack skills or vocabulary), leading group oral or silent reading and review, and administering tests.

Discussion and social interaction with pupils. This category indexes a somewhat different type of teacher-pupil interaction. Discussion and social interaction involve more extended pupil talk, and pupils comments are not generally restricted to reading aloud and answering specific questions from the teacher. The pupils interact with teacher on a much higher level of creative or interpretiv thought than in interactive roles.

KI

Teacher assisting and monitoring pupil work (facilitative). In these roles, the teacher provides support and assistance to pupils who are engaged in

relatively independent activities. The teacher is facilitating assigned seatwork (reading stories, using workbooks or dittos; etc.) or self-instructional activities (instructional games, audiovisual equipment, or creative work).

Teacher idle. The teacher has been coded as doing nothing for a period of .

at least one minute during the ten-minute observation episode.

Table 2

Relative Contributions of Teacher Roles to Instructional Mode Scales

Name of Teacher Mode	Major Contributors	Other Contributors
Teacher directive roles	Managing (51%) Assigning task (32%) Instructing (10%)	Story telling, reading aloud (3%) Disciplining (2%) Reciting poetry) less Interrupted by office) than Praising) 1%
Teacher-initiated i interactive roles	Drilling (42%) Oral/Silent reading and review (42%) Testing, assessing (12%)	Tutoring (5%)
Discussion and social interaction with pupils	Discussion (66%) Social interaction (34%)	
Teacher assisting and monitoring pupil work	Facilitating pupil seatwork (85%) Facilitate self-instruction activities (15%)	תכ
Teacher idle	Doing nothing (100%)	

A profile reflecting the proportion of a given teacher's time in these five modes was generated by calculating the proportion of time in each mode within each observation episode then taking the average across all episodes for the teacher. Mean proportions of time in these modes, averaged across all 56 teachers, are displayed in Table _3. The teacher-initiated interactive mode clearly dominates, accounting for an average of 70% of teacher time. The directive and facilitative modes together represent an average of 25% of teacher time. Discussion and social interaction are comparatively rare, accounting for an average of less than half a minute per episode.

2i

Table 3

Average Proportion of Teacher Time in Five Activity Modes

	Proportion of Time in=Mode		
Activity Mode	Mean	Standard Deviation	
Directive	14.43%	10.665%	
Teacher-initiated interactive	70.28	21.235	
Discussion and social interaction	3.25	6.590	
Assisting and monitoring pupil work	10.63	13.111	
Idle	1.41	3.237	

during Teacher-Focused Observations

Pupil roles were grouped into four nonoverlapping participation modes which generally correspond to the teacher modes. The roles in each mode, and their relative prominence within the mode, are displayed in Table 4. The four pupil modes or levels of activity are:

<u>Pupils receiving directions</u>. In these roles no immediate verbal response or activity (other than listening) is required of pupils. These roles are most often observed in conjunction with teacher directive roles, as pupils listen to the teacher make an assignment, carry out classroom management tasks, or lecture.

<u>Pupils responding to teacher</u>. Pupils are involved in activities that are led by the teacher but that call for them to respond, either as individuals or in a group. These roles are the pupil counterparts to the teacher-initiated interactive roles--drill, oral/silent reading and review, and testing.

<u>Pupils engaged in seatwork and self-instruction</u>. These roles call for the most active level of participation from pupils. Most often pupils are working fairly independently, completing seatwork assignments or carrying out self-instructional activities (working with audiovisual

22

equipment or instructional games, working on creative tasks. Less often pupils are involved in discussions or social interactions.

<u>Pupils idle</u>. Pupils have been coded as in transition (waiting to begin a new task or to get the teacher's attention) or as not attending to task for at least one minute.

Relative Contributions of Pupil Roles to Instructional Mode Scales

Name of Pupil Mode	Major Contributors	Other Contributors
Pupils receiving directions	Being managed (42%) Being assigned task (38%) Being instructed (12%)	Listening to story (5%) Being disciplined (3%) Being praised (less than 1%)
Pupils responding to teacher	Oral/silent reading and review (45%) Drill (40%) Being tested, assessed (11%)	Being tutored (5%)
Pupils engaged in seatwork and self-instruction	Seatwork (64%) Self instructionAV games, creative work (19%)	Discussion (13%) Social interaction (3%) Clean-up)less Reciting poems)than Visit resource ctr) 1%
Pupils idle	In transition (65%) Not attending to task (35%)	

A profile of pupil time in each mode was generated for each classroom using essentially the same procedure followed in generating the teacher profiles. Proportions of pupil time in the four modes is presented in Table <u>5</u>. Pupils working with the teacher during reading instruction spend an average of 65% of their time in the responding mode.

Discussion and social interaction were included in this category rather than treated separately because, like seatwork and self-instruction, they were assumed to involve highly active participation of pupils. For teachers, discussion and social interaction were treated separately in order to examine differences in instructional style. These two roles were rare for both teachers and pupils.

23

Table 4 •

Table 5

Proportio Mean	on of Time in-Mode Standard Deviation	
9.02%	18.209%	,
65.10	20.128	• •
•		
20.35	7.202	•
5.52	5.764	
	Mean 9.02% 65.10 20.35	9.02% 18.209% 65.10 20.128 20.35 7.202

Average Proportion of Pupil Time in Four Activity Modes during Teacher-Focused Observations

Most of their remaining time is spent completing seatwork assignments or working on self-instructional activities. On the average, comparatively little pupil time is spent receiving directions. The low proportion of idle time among pupils may be due in part to the observers' focus on only those pupils who were working with the teacher.

The relative proportions of a given teacher's time in the five teacher activity modes were used as a profile of that teacher's instructional style. The pattern of pupil time use within the teacher's classroom has been treated as a second profile, reflecting how pupils in the classroom experience reading instruction. A comparison of the teacher and pupil profiles shows strong similarities in the relative proportion of time spent in corresponding teacher-pupil modes. The correlation between use of pupils and teachers time (presented in Table 6.) indicates these strong similarities. Correlations between the time the teacher spent in the directive, interactive, and facilitative modes and the time the pupils spent in the counterparts of those modes (i.e., receiving directions, responding to teacher, and seatwork/self-instruction, respectively) are high and positive. Correlations for time spent in noncorresponding modes are generally negative.

24

In another section of the instrument, observers recorded information about grouping patterns and pupil activities for the classroom as a whole. Data from these observations showed a higher frequency of *down time* among pupils not working with the teacher or an air

Perferct correlations would be obtained only if all pupils in the teacher's group were engaged in a single type of activity, corresponding to the teacher's activity, and pupils were never idle unless the teacher was also idle.

Correlations between Proportions of Teacher and Pupil Time

Corresponding and Noncorresponding Modes

during Teacher-focused Observations

÷≓

• • • • • • • • • • • • • • • • • • •	· · · · · · · · · · · · · · · · · · ·	Teach	er Mode	
Pupil Mode	Directive	Interactive	Facilitative	İdle
Receiving directions	1.71	31	08	.00
Responding to teacher	49	.94	64	40
Seatwork, self-instruction	.08	80	.78	.32
Idle	.58	38	12	.36] *

Profiles of teacher and pupil time use by treatment group and grade level are displayed in Table 7. Reliable differences across treatment groups are found in both the teacher and pupil profiles.

Teachers in the saturated classes spent 70% of the time in the interactive mode. Correspondingly, the pupils in those classes spent 72% of the time in the interactive mode. The teachers in the concentrated classes, although spending a majority of the time in the interactive mode, used seatwork and self-instruction activities for their pupils ° significantly more often than the teachers in the saturated classes. Consequently, the teachers in the concentrated classes spent more time monitoring or assisting their pupils in these activities than the teachers in the saturated classes. Pupils in the concentrate classes spent an average of 28% of the time in seatwork and self-instruction. This represents almost twice the amount spent by pupils in the saturated classes.



Average Proportions of Teacher and Pupil Time in Instructional Modes

during Teacher-Focused Observations by Treatment Group and Grade

		ge Proport:	ion of Tim	e in Mode		Group	•
	Satur		Concen	trated	Sigr	if. at	1
Instructional Mode	Grade 2	Grade 4	Grade 2	Grade 4	<u> </u>	05	/
Teachers	•			••	•	. /	
Directive	12.28%	14.27%	15.00%	16.00%		ەر	•
Intèractive	78.26	77.06	69.04	56.15		X	
Disc. & soc. interaction	1.63	[:] 3.07	3.83	4.46			שי ז
Assisting & monitoring	7.28	. 4.28	12.13	19.37		X	, t
Idle	0.54	1.32	0.00	3.91		14 12	
Pupils							45
Receiving directions	6.88	7,99	12.33	9.25			
Responding to teacher	74.23	69.58	66.64	49.41		X	
Seatwork & self-instruct.	12.40	15.69	17.65	36.02	•	X	
Idle	6.52	6.64	3.38	5.33			•
(Number of Cases)	(15)	(14)	(13)	(14)	-		

Likewise, teachers in concentrated classes spend a significantly greater amount of time (16%) facilitating these activities than do teachers in saturated classes (6%).

Examination of grade-level differences within treatment group reveals an additional pattern. In saturated classes, the profiles are very similar in both grades. However, the concentrated classes' profiles apparently differed from second to fourth grade. Specifically, the diversity of teacher and pupil activities increases from second to fourth grade. Fourth-grade pupils in concentrated classes spend more than a third of their time in seatwork and self-instruction--twice as much time as the second graders. Likewise, fourth-grade teachers in concentrated classes spend more of their time assisting pupils than any other group of teachers.

22.

The data in Table <u>7</u> indicates that the teachers in saturated classes relied almost exclusively on the interactive mode. The instructional style of the teachers in the concentrated classes is more diverse, thereby allowing pupils to experience a wider range of activities in teacher-led reading groups. This diversity is more pronounced in the fourth grade. One possible explanation is that teachers in concentrated classes use various modes of instruction as a technique for concentrating services on EDY pupils. Unfortunately, data from the individual observation instrument (which will be discussed in a subsequent section) do not show systematic differences in use of time by EDY and nonEDY pupils in the concentrated classes.

The use of aides in the classroom contributed to the differences between the treatment group's instructional profiles. Aides were used more frequently in the saturated classes. Data from the whole-class indicated that aides frequently assisted and monitored pupils in seatwork and self-instruction activities (the whole-class observation was made during ten-minute break between the two teacher-focused observations When an aide was available, pupils using audiovisual equipment and games or completing their assigned seatwork apparently worked unsupervised or with the aide, rather than with the teacher. Consequently, seatwork and self-instruction were rarely observed in the teacher-focused observations when an aide was present. In classrooms without aides, these activities were generally pursued within the context of the teacher's group, and therefore were more likely to be included in the teacher-focused observations.

Data from the individual observations, which were not limited to pupils working with the teacher showed that pupils in saturated classes spent more time in seatwork and self-instruction than was indicated by the teacher-focused observations.

The individual data do not indicate that concentrated teachers vary instructional modes in the same way for all pupils, jonly that variations in mode are not reliably associated with the child's EDY status.

ERIC FullTaxt Provided by ERIC

However, the data from the individual observation instrument reaffirmed that pupils in the concentrated clases spent significantly more time involved in seatwork and selfinstruction than the pupils in the saturated classes. Therefore, using more aides in saturated classes apparently accounts for only part of the observed differences in instructional style across treatment groups.

Instructional Materials

The observers also recorded information concerning the types and diversity of instructional materials used during the teacher-focused observations. The primary materials used were textbooks (used during about half the teacher observations) and workbooks (about 38% of the observations). Other materials include blackboard or magic slate (21%), paper and pencil (16%), flashcards or teacher-made materials (13%), and dittos (11%).

The average number of different materials used during a ten-minute episode was 2.00. An average of 2.20 materials per episode were used in the concentrated classes, which is significantly higher than the average of 1.86 materials used in the saturated classes ($p_{<}.05$). In addition, teachers in concentrated classes were more likely to use materials purchased with EDY funds. Approximately 43% of the materials used in concentrated classes were purchased with EDY funds. Teachers in saturated classes used approximately 27% EDY materials ($p_{<}05$). Although no reliable differences in overall materials usage were found across grade levels, fourth-grade teachers in both treatment groups made proportionately greater use of EDY materials than secondgrade teachers.

Teachers' Interpersonal Style

The teachers' interpersonal style was also observed during the teacher-focused observations. Observers recorded both the frequency and the intensity of the following teacher behaviors:

24.

 $2\overline{3}$

Supportive verbal expression--comments from the teacher praising pupil work or behavior.

Supportive nonverbal expression -- actions by the teacher indicating approval of pupil work or behavior, ranging from smiling or putting stars on pupils work to putting an arm around a pupil. _-

Nonsupportive verbal expression--comments from the teacher criticizing or showing disapproval of pupil work or behavior.

<u>Nonsupportive nonverbal expression</u>--actions by the teacher indicating disapproval or criticism of pupil work or behavior, such as frowning, distributing markers of poor performance, or making a list of disruptive pupils.

Behavioral data for each of the 56 teachers were combined across observations to form several indices of the teacher's interpersonal style or responsiveness. The measures are described in Table 8, which also shows the mean score across all 56 teachers for each scale. Fundamentally, scales were developed on three levels. On the first level, scores on verbal and nonverbal responses were combined, retaining both the distinctions between frequency and intensity of these responses and the distrinction between supportive and nonsupportive teacher behaviors. On the second level, frequency and intensity were combined to generate separate measures of supportive and nonsupportive affect. On the third level, the supportive and nonsupportive scales were combined to provide an overall index of the teacher's responsiveness or interpersonal style.

The mean scores presented in Table <u>8</u> suggest two general findings: First, even though virtually all teachers displayed some supportive affect, the display was relatively infrequent and at a low level of intensity. The majority of teachers took care to praise the students only when appropriate. Second, the vast majority of teachers rarely commented or acted in a manner which indicated disapproval of pupils' work or

"Statement such as "Yes, that's right" or "No, that's wrong" were regarded as neutral feedback and were not counted as instances of supportive or nonsupportive verbal expression. Only comments that included praise or criticism were counted.

25.

Table 8

Teachers' Interpersonal Style Variables Created from Data Collected

during Teacher-Focused Observations

Interpersonal Style Variables	Contents of Variables	- Score Bange	Mean Score (n=56)
First-level variables			
Frequency of supportive responses	Frequency of supportive verbal + nonverbal	2-8	3.95
Intensity of supportive responses	Intensity of supportive verbal + nonverbal	2-8	3.80
Frequency of nonsupportive responses	Frequency of nonsupportive verbal + nonverbal	2-8	2.13
Intensity of nonsupportive responses	Intensity of nonsupportive verbal + nonverbal	2-8	2.15
Second-level variables			
Supportive affect	Freq of supp X intensity of supp	4-64	18.65
Nonsupportive affect	Freq of nonsupp X inten- sity of nonsupp	4-64	4.83
Third-level variable			and the state
Teacher responsiveness	Supportive affect X nonsupportive affect	16-409	6 88.21

behavior. No criticism or disapproval was observed during any of the eight ten-minute episodes for 52% of the teachers. Only 11% of the teachers averaged more than one instance of criticism or disapproval per episode.

Correlative analysis of the affect variables revealed additional general findings regarding teachers' interpersonal style: Teacher praise and approval are not related to criticism and disapproval (at least for this group of teachers). Essentially, teachers who score relatively high on the positive measures are neither more nor less likely than other teachers in the sample to score high on the negative (nonsupportive) measures. Supportive and nonsupportive responsiveness apparently function as relatively independent components of these teachers' interpersonal styles.



26.

Mean scores on the responsiveness scales are displayed by grade level within treatment group in Table 9. Teachers in concentrated classes made more frequent support ive affect than teachers in saturated classes. (Only the intensity measure reaches the criterion significance level of .05 for treatment group differences; however, difference between teachers in saturated classes and teachers in concentrated classes on both the frequency scale and the overall measure of supportive affect are in the same direction and approach the criterion significance level.) Reliable differences across treatment groups were not found for the nonsupportive scales or for the overall responsiveness ind

Table 9

	Av	Tmt Group Differen ce				
	Saturat	ed	Concentrated		Signif at	
Variable	Grade 2	Grade 4	Grade 2	Grade 4	P<.05	
Frequency of supportive responses	3.56	2.92	4.12	3.44		
Intensity of supportive responses	3.51	2.86	4.04	3.46	x	
Frequency of nonsupportive responses	2.44	2.23	2.35	2.20	· ·	
Intensity of nonsupportive responses	2.49	2.25	2.47	2.27		
Supportive affect	15.03	9.40	19.26	13.93		
Acnsupportive affect	7.04	5.75	6.42	5.35		
Teacher responsiveness	116.05	47.41	121.98	78.97	-	
(Number of cases)	(15)	(14)	(13)	(14)	ਕ	

Teachers' Interpersonal Style by Treatment Group and Grade Level

In both treatment groups, the second-grade teachers consistently received higher scores than the fourth grade teachers on the supportive and nonsupportive measures. This indicates that the second grade teachers used immediate praise or disapproval more frequently than the fourth grade teachers (disapproval was relatively rare response towards pupils). Therefore, with the sample of second and fourth grade teachers, the interpersonal style was found to be related more to grade than to treatment group, with teachers responding more frequently to second grade pupils.

34

Individual Pupil Observations

The Individual Student Instrument was designed to focus on specific students, selected in advance, to obtain information about their activities, use of materials and shifts in group involvement during the entire reading period. The observer focused on individual students who were pre-selected according to $g\overline{r}$ de level, EDY status and the type of class treatment in which the students were involved. Each of four students per classroom was observed for approximately 30 minutes.

The individual pupil observations were designed to obtain information describing instructional resources and processes used by teachers in the two treatment conditions. Unlike the teacher-focused observations, the focus on individual pupils provided additional information about the implementation of the saturated or concentrated treatment - i.e., how teachers differentiate resources and processes used on the basis of pupil EDY status. The additional information was used to address the basic issue of whether teachers used different materials and/or methods with EDY student than those used with nonEDY students, and if so, under what treatment conditions did such differentiation occur. The use of different methods and materials for EDY and nonEDY is assumed to be consistent with implementation of the concentrated treatment, providing that the use reflects the focusing of EDY resources and services on EDY pupils. In saturated classes, however, differences between EDY and nonEDY would not necessarily be expected. The individual instrument was structured around instructional activities in which the target pupil was engaged over the observation period. An activity is defined as including four elements:

Pupil role. The nature of the task in which the child is engaged (e.g., drill, oral/silent reading, receiving assignment)

<u>Materials usage</u>. The type(s) of material with which the child is working (e.g., textbook, workbook, blackboard, dittos); the source of funds used to purchase each material being used (EDY, other, combination of EDY and other funds).

<u>Group leadership</u>. The instructional leader(s) of the group in which the child is working (teacher, aide, other adult, cross-age tutor); activities not led by an adult or cross-age tutor are defined as self-directed.

Group size. The total number of pupils in the group of which the target child is a member.⁺

Whenever one or more of these elements changed, the current activity was considered complete and a new activity (combination of pupil role, materials, group leadership, and group size) was recorded. Observers also recorded the number of minutes associated with each activity in which target children were engaged.

The pupil role codes developed for the Classroom Observation Instrument were also used in the individual observations.

A group is defined as two or more pupils working with the same group leader and/or involved in the same activity (with or without a leader) and arranged in close physical proximity to each other. Pupils working on a common task but not seated together were considered to constitute a group only if a teacher or other leader identified them as such.

33

A two-stage sampling procedure was used to befect pupils for individual observations on the basis of three sampling variables: treatment condition (first stage), grade level (first stage) and EDY classification (second stage). The two first-stage variables were determined at the classroom level; that is, four pupils were selected from each of the 72 classrooms representing the four treatment group/grade level permutations. Within each classroom, individual pupils were chosen on the basis of their EDY classification. Whenever possible, EDY pupils were selected from the lowest quartile (first to twenty-fifth percentile) and nonEDY pupils from the highest quartile (seventy-sixth to ninety-ninth percentile). This procedure facilitated the determination of whether instructional patterns (resource utilization, mode of instruction, etc.) differed for EDY and nonEDY children in either treatment condition. The assumption was made that whatever distinctions a teacher made between EDY and nonEDY pupils would be most apparent by focusing on the individual observations on very low-scoring EDY pupils and comparatively high-scoring nonEDY pupils.

SARIATE

V E L

The three sampling variables constitute an eight-cell matrix (see Figure 1), with 36 children in each cell. Within each cell children were selected to reflect as closely as possible the ethnic composition of the overall 72-classroom population (approximately 58% Spanish surname, 23% caucasian, 12% black and 7% other) and to include equal numbers of boys and girls.

-	Concentrated	Saturated
GRADE 2	EDY (36)	EDY (36
<u></u>	nonEDY (36)	nonEDY (36)
GRADE 4	EDY (-36)	EDY (36
	nonEDY (36)	nonEDY (36)

TREATMENT CONDITIONS

Figure 1

34

Target Sample for the Individual Pupil Observations

EDY status was determined by Total Reading score obtained on the MAT administered in Fall 1977. Children scoring the 50th percentile (according to the national norms developed by the publisher) were percentile were classified as nonEDY.

The results reported here are based on the 219 pupils with individual observation data and pre- and post-scores on the MAT. These 219 children constitute the analysis file for the integrated analyses of pupil-level data reported in the analysis section. Demographic characteristics of the 219 pupils are displayed in Tables 10 and 11.

Table 10

Demographic Characteristics of Pupils in the Individual Pupil File: Second Grade Subset (N=118)

Concentrated (63)		Saturated (55)		
EDY (32)	nonEDY (31)	EDY (32)	nonEDY (23)	
21	17	21	31	
7	9	6	· 8	•
· 3	1 •	5	2	
1	4	-	-	
	EDY (32)	EDY (32) nonEDY (31)	EDY (32) nonEDY (31) EDY (32) 21 17 21 7 9 6	EDY (32) nonEDY (31) EDY (32) nonEDY (23) 21 17 21 31 7 9 6 8

Table 11

Demographic Characteristics of Pupils in the Individual Pupil File: Fourth Grade Súbset (N=101)

	<u> Concent</u>	trated (51)	Saturated (50)		
· · · · · · · · · · · · · · · · · · ·	EDY (29)	nonEDY (22)	EDY (25)	nonEDY (24)	
Spanish surname	16	10.	15	14 🧭	
Caucasian	7	7	4	8	
Black	3	3	. 4	2	
Other	. 1	2	2	_	



'

31.

Individual Pupil Observations--Descriptive Results

Engaged Time

A measure of engaged time was generated for each of the 219 pupils by calculating the total number of minutes the child spent in all roles classified as instructional (see Appendix A.2). The results, displayed in Table 12. are very similar to those obtained from the analysis of the teacher-focused observations. Specifically, second- and fourth-grade pupils spent an average of 87% and 90% of their time, respectively, engaged in activities directly related to instruction. Reliable differences between treatment groups were not found at either grade level. EDY and nonEDY students in both saturated and concentrated classes generally had a high proportion of engaged time. In only one EDY-nonEDY comparison is there a significant difference in engaged time. In the fourth-grade concentrated classes, EDY pupils spent an average of 95% of their time in instructional roles, while the nonEDY pupils spent an average of 85% of theirs in instructional roles. This difference (largely due to the differences between these two groups in average proportion of idle time) accounts for about three minutes during the half-hour observation period. The average engaged time did not fall below 25 out of the 30 minutes for any of the groups.

15

Table 12

	a	Average Proportion of Engaged Time						,
		Conce EDY	nonEDY	<u>Satur</u> EDY n		A11 Concentrated	All Saturated	All Pupils
Grade 2		85.2%	89.7%	89.1%	83.7%	87.5% _	86.8%	87.2%
Grade 4	2	95.2	85.7	90.2	88.9	91.2	89.2	90.2

· 32. 36

Pupil Engaged Time during Individual Pupil Observations

ERIC[®]

Instructional Modes

To examine relationships between EDY status and instructional processes, pupil roles were grouped into the four previously discussed activity modes (receiving directions, responding to teacher or other group leader, seatwork and selfinstruction, and idle--see Table <u>4</u> above). The average proportions of time spent in these modes by second and fourth grade pupils are displayed in Table <u>13</u> and Table <u>14</u>. Generally, the results are similar to the results obtained from the teacher-focused observations, with the excéption that time spent in seatwork and self-instructional activities is proportionately higher in the individual data.

Second grade teachers in the concentrated classes apparently made some distinctions between EDY and nonEDY pupils. EDY pupils spend somewhat more time than nonEDY in the responding mode (drill, oral/silent reading, etc.). Moreover, nonEDY children spend more time receiving directions (e.g., being given assignments) and completing seatwork and self-instruction tasks. Within the saturated classes, the majority of time was fairly evenly divided between the responding mode and the seatwork/self-instruction mode for both EDY and nonEDY pupils.

The results from the fourth grade differ. Fourth grade teachers from neither treatment groups made systematically different use of the three instructional modes for EDY and nonEDY. Within the fourth grade, patterns were very similar within treatment type but very different between treatment type. Specifically, pupils in concentrated classes spent an average of half their time in seatwork and selfinstruction (the responding (interacting) mode no longer dominates). However, in saturated classes, the teachers generally use the interactive mode for EDY and • nonEDY children alike.

Full fact Provided by ERIC

33.

Table 13

Average Proportion of Pupil Time in Four Activity Modes during Individual Pupil Observations:

Ξ٠

Second Grade (N=118)

	A	verage Pr	oportic	on of Tim	e in Mod	e		Tmt. Grp Diff.
	Conc	entrated nonEDY	Satur EDY	nonEDY	All Conc.	All Sat.	All Pupils	Signif. (p<.05)
Receiving directions	5.6%		5.9%	3.97	5.5%	5.0%		x
Responding	52.6	43.4	45.6	39.4	47.9	43.0	45.7	
Seatwork/self-instruct.	29.1	36.3	42.4	42.1	32.8	42.3	37.1	
Idle	12.1	8.2	7.7	14.2	10.4	10.4	10.4	
•		· .						

Table 14

Average Proportion of Pupil Time in Four Activity Modes

during Individual Pupil Observations:

Fourth Grade (N=101)

	Average Proportion of Time in Mode							Tmt. Grp Diff.
		nonEDY		nonEDY	All Conc.	All Sat.	All Pupils	Signif. (p<.05)
Receiving directions	5.8%	5.0%	5.9%	3.9%	5,5%	5.0%	5.2%	
Responding	34.3	35.5	62.2	61.0	34.8	60.4	47.4	x
Seatwork/self-instruct.	57.4	48.7	24.4	26.0	53.8	26.1	40.2	x
Idle	2.4	10.8	7.4	9.0	5.9	8.5	7.2	



34.

-33

In the concentrated classes, results from the individual students observations are similar to the findings from the teacher-focused observations. Within the second grade classes, teacher- (or other group leader) initiated interactions represented the primary instructional mode, particularly for EDY pupils; however, teachers frequently used independent seatwork activities as a secondary instructional mode, especially for nonEDY pupils. Within the fourth grade classes, pupils did more, although the interactive question-response mode still accounts for about a third of EDY and nonEDY pupils' time.

The pattern of time use for fourth grade pupils in saturated classes is consistent with findings from the teacher-focused observations. Group interaction with the teacher or other leader dominated pupils were involved in seatwork about 25% of the time--half as much as in concentrated fourth-grade classrooms. Findings from the second-grade saturated group observations are less consistent with the teacher-focused observations. Seatwork and self-instruction, which accounted for comparatively little time in the teacher-focused observations conducted in second grade saturated classes, are relatively prominent in the individual data. This is probably because second grade teachers in the saturated classes generally used seatwork and self-instruction for pupils outside the group with which the teacher was working.

N.

Instructional Materials

Patterns of materials usage recorded during the individual observations are similar to the patterns found in the teacher-focused data. Pupils used textbooks and workbooks most often; however, pupils commonly used paper and pencil, dittos, and blackboard.

Diverse use of materials apparently does not differ as a function of grade level, treatment condition, or EDY classification. However, fourth grade pupils in concentrated classes used EDY materials more often than fourth grade pupils in the saturated classes. EDY materials were directed mainly to EDY pupils in these fourth grade concentrated classes (see Table 15). Proportional use of EDY materials was at least two times greater among EDY pupils in concentrated classes than among any other

3.)

group of fourth graders. Fourth grade EOY pupils used EDY materials proportionally greater than any group of second graders. This finding is consistent with implementation of the concentrated treatment.

Table 15

Proportion of EDY materials Used during Individual Pupil Observations

	Ϋ́.		Average Proportion of EDY Materials Used						
	:	Concen EDY	nonEDY	Satur EDY	nonEDY	All Conc.	All Sat	All Pupils	Differenc Signif. (p<.05)
Grade 2		23.5%	29.5%	15.0%	29.7%	26,5%	21.2%	24.0%	÷
Grade 4	Ì	48.9	25.2	15.0	22,4	38.7	18.7	28.7	X

Group Size

1-

Average group size was about 10 in both the second and fourth-grade classes. Group size did not differ reliably between treatment groups in either grade. In the concentrated classes, groupings do not appear to differ between EDY and nonEDY pupils. In saturated second-grade classes, however, nonEDY pupils generally worked in smaller groups than EDY pupils. This distinction was not found in the fourth grade saturated classes.

Table 16

Average Size of Instructional Groups during Individual Pupil Observations

	•	Average Size of Instructional Group							
1	Conce	Concentrated		Saturated		A11	A11	Difference Signif.	
	EDY	nonEDY	EDY	nonEDY	Conc.	Sat.	Pupils	(p<.05)	
Grade 2	10:4	11.4	11.0	8.0	10.9	. 9.7	10.3		
Grade 4	9.7	11.1	9.3	10.3	10.3	10.0	10.1	•	

 $\sqrt{40}$

Two basic questions guided the analysis of study data for evidence of treatment

effects:

- (1) Does saturation or concentration of compensatory resources and services--to the extent such occurred in this study--relate to reliable and meaningful differences in basic reading skill attainment?
- (2) What are the contextual and procedural (instructional) conditions which account for differences in reading skill attainment?

The first question focuses on the effectiveness of the implementation of the two treatments. Given the administrative, logistic and financial aspects of saturation vs. concentration of treatment, the essential objective was to determine which method '(treatment) of dispensing compensatory educational resources and services should be utilized to attain highest pupil reading aptitude. An additional objective was to determine the central and peripheral effects of the allocations of these treatments on both class practices and pupil learning (reading, as measured by MAT). Throughout the analysis, an awareness of additional issues was essential to determine whether they should be included in this study or further studies.

The second analysis question focuses on the more general domain of instructional effects. The primary objective here was to determine, from evidence gathered in this study, which characteristics of pupils, resources, and instructional procendres taken together, accounted for learning outcomes (i.e., reading skills, as measured by the MAT). This "input-process-outcome" analysis represents an empirical extension of the emerging teacher effectiveness research and provides a basis for policy alternatives to the saturation-concentration intervention being investigated. In other words, how do these findings compare with those from other major studies of teacher effectiveness (e.g., the Beginning Teacher Effects Study), and how these findings might include alternative interventions for improvement of reading skills?

The remainder of this section is divided into four parts. Part I contains a discussion about the development of analysis variables. Part II contains an examination of the data for evidence of effects due to saturation or concentration, using data

37:

collected at the class-level. Results of these analyses are reported separately for the second and fourth grades. Part III contains an extension of the analysis to include information on the relative effects of the two alternative modes of delivering compensatory resources and services at the individual pupil level, using evidence gathered from the sample of 219 children. This provides an examination of pre-post test patterns in terms of degree of educational disadvantage, ethnicity, gender, and the interactions of these conditions with the alternative "treatments" as implemented by the teachers. Finally, Part IV deals with the more general question of how this information regarding contextual and instructional processes used in the class explain outcomes observed at the class level.

Development of Analysis Variables

These analyses have not been carried out as a set of hypothesis-testing activities. Although statistical tests of probability were used in the effects of saturation vs. concentration, greater emphasis was placed on identifying and better understanding the proximal (near) and distal (far) consequences of this attempted intervention. Additional emphasis was placed on evaluating such effects against alternative input-process-outcome patterns detected in the data. Variable's designed to accomplish these analyses were derived from both consideration of the fundamental issues in the demonstration project (i.e., saturation vs. concentration of resources and services) and from an awareness of the results of contemporary teacher effectiveness research. Therefore, two criteria were employed in defining and developing variables for formal analysis of the data:

- A logical or manifest relationship to the demonstration project's goals and objectives (referred to as "implementation" variables); and/or
- (2) A logical or manifest relationship to constructs identified as important in contemporary teacher effectiveness research.

Variables developed from either criteria can be analyzed as input, process, or outcome indicators. Consequently, a measure of group size (pupil-teacher ratio) could be used as either an outcome indicator in an analysis of treatment implementation, or an input or process measure in an analysis of determinants of variation in reading scores.

A brief summary of the variables or constructs developed for data analysis, including the source, operational definition, and descriptive statistics, is presented in Appendix B.

Moreover, the analysis of each data source was performed both independent of and concurrent with all other data sources in attempting to identify the optimal reduced set of variables for formal statistical treatment. The final set of analysis variables and their descriptive parameters are summarized in Appendix B. Since

39.

these are data sets for integrated analyses (at the teacher and pupil levels, respectively), the requirement that each case be based on complete data from all sources resulted in a reduction of approximately 20% in the overall data base (i.e., from 72 to 56 teachers, and from 244 to 219 pupils).

Since measurements are aggregated over multiple observations (usually eight ten-minute observation episodes for teacher/class variables, and up to 16 "roles" for individual pupil observations), both the mean (central tendency) and standard deviation (variability) for each variable are employed in subsequent analyses. This provides a basis for indirectly assessing the relative importance of stability and variability of process variables in accounting for outcome variance. For example, "individualization" as an instructional technique would imply higher standard deviations on process measures over the eight observations than would "routinization" as a technique. Although the converse of the previous statement is not logically sufficient (i.e., high standard deviations on process measures do not themselves provide a sufficient condition to conclude that a teacher is "individualizing"), the inclusion of both moments (means and standard deviations) is useful in developing a better understanding of complex instructional processes and their relationships to criterion patterns.

40.

II. Analyses of the Effects at the Teacher/Class Level

The information obtained either during the classroom observations, through interviews and test data, or through additional methods was scrutinized to determine whether complete data on all relevant measures were available for each class.^{*} The requisite complete data set was found for 56 of the teacher/class units observed. The teacher/class unit was the elementary unit of analysis; therefore, any data not specifically measured on the level of the teacher/class unit was aggregated into the data base for the teacher/class unit as follows:

- (1) Teacher interview variables remained unmodified.
- (2) Role descriptors gathered for individual observation procedures were aggregated to the classroom level, and appropriate statistics (mean and standard deviation) were computed.
- (3) Means and standard deviations were computed across successive observation episodes for each candidate variable.
- (4) Principal interview variables were imputed to teachers within their respective schools.
- (5) Average classroom compositional and performance (MAT score) indicators were calculated for current year and prior year classes for teacher.

The resultant data set, showing mean values (and standard deviations, where relevant) on each of the final analysis variables, is presented in Appendix B.1. (See Appendix C for intercorrelations among these variables.

Relevant measures are defined as those showing a both substantial (p<.2) and a nonoverlapping relationship either to the treatment variable; therefore, approximating and implementation variable and/or to the outcome measures.

The alternative procedure of combining prior year scores of current pupils was rejected because of a variety of technical and analytic considerations, such as pupil mobility and attrition (ranging from 20-85%), different pretest form, nonheterogeneous prior cla assignments and prior research on the stability of teacher effects. Moreover, the procedure adopted for this analysis better assures identification of instructional trai among "truly" effective teachers (i.e., those who consistently produce high-scoring pup rather than focusing on instructional effects at individual pupil level. The latter is is addressed by the individual pupil analysis, the results of which are presented in Part II of this chapter.

43

Two types of conditional analysis were subsequently performed on these data. Both types are based on the general linear hypothesis.

In the first analyses, two-way analyses of covariance were performed on each of the four outcome variables (total reading, word knowledge subscore, word analysis subscore, and reading subscore) within grade level (second and fourth). The treatment condition (saturated or concentrated) was used as the between-group variable. The five context or process measures which showed the closest relationship were treated as covariables.

In the second analysis, multiple linear regressions were performed on these data. The mean reading achievement scores were regressed on several combinations of context and process variables to identify the most significant determinants of outcom score variance. Through this analytical technique, an assessment of the instructional effects of several process variables was possible when the effects of context variables were sufficiently controlled. Put more succinctly, once the class composition was controlled, the process (or instructional) variables that accounted for differences in reading achievement were identifiable. The "process" determinants identified by these multiple regression techniques were used as covariables in the analysis of covariance. The findings from the multiple regressions are reported in Part IV.

46

Table 16

	Unadjust ment l	ted Treat- Means	43	Adjust	ed-Means	•	
Outcome Measure	Conc. (n=13)	Sat. (n=15)	Significance of diff (~)	Conc.	- Sat.	Beta	2^2
Total Reading	49.4	49.8	0.79	49.7	59.6	• 0.01	.35
Word Knowledge	52.8	52.4	0.76		52.7	0.04	.2]
Word Analysis	48.2	48.4	0.83	47.9	48.6	0.12	.3;
Reading Subtest	50.2	49.8	0.89	49.4	50.4	0.12	.2!
Average	50.1	50.1	0.82	49.8	50.3	0.07	.2'

Outcome Score Analysis for Grade 2 Classes (Class-level Data)

Results of Covariance on the 28 Second Grade Classes

The results of the classroom-level analysis of effects within the second grade subsample (28 classes) are summarized in Table <u>16</u>. Clearly, none of the observed treatment effects are statistically reliable. Less than one-third of the criterion variance (average R^2 =.296; maximum=.37; minimum=.21) is demonstrated even when context and process covariables are included. The magnitude and direction of the observed and adjusted mean differences for the second grade subsample clearly indicate the absence of reliable effects. In fact, the average of observed means across the four outcome measures is virtually identical across the two treatments (50.1), and nearly identical for adjusted means (49.8 for concentrated classes, 50.3 for saturated classes). Moreover, as indicated in a subsequent section of this study, the context and process covariables usually did not account for much additional criterion variance within this second grade sample.

Outcome Score	Analysis	for	Grade	4	Classes	(Class-level	Data))
---------------	----------	-----	-------	---	---------	--------------	-------	---

	Unadjust ment N	ted Treat- Means		Adjuste	d Means	, <i>.</i>	~
Outcome Measure	Conc. (n=14)	Sat. (n=14)	Significance of Diff (~)	Conc.	Sat.	Beta	R ²
Total Reading	65.3	61.6	0.05	65.2	<u> </u>	0.29	.56!
Word Knowledge	67.6	63.4	`0 .02	67.5	63.5	0.33	.64:
Word Analysis	(n	ot ap	plicable)				
Reading Subtest	64.6	61.4	0.09	64.7	61.3	0.27	.541

Results of Analysis of Covariance on the 28 Grade 4 Classes

The results of the class-level analysis of effects within the fourth grade subsample (28 classes) are summarized in Table 17. These results can be interpreted as follows:

- (a) For both the total reading and the word knowledge measures, concentrated services produced reliably greater mean scores than did saturated services. These effects were evident both before and after adjustment for process-context covariates (which include prescores).
- (b) Mean differences on the reading subscore (basically a reading comprehension subtest) favor the concentrated condition (p<.09).
- c) The magnitude of these mean differences averages approximately four standard score points, or about 15 percentile points (based on the MAT equipercentile scale). Specifically, based on national norms, the approximate percentile equivalents of the fourth-grade average scores are:

	Adjusted Mean Score				
	Concentrated				
Total Reading	38	24			
Word Knowledge	40	24			
Reading Comprehension	36	26			

43

III. Results from Analyses of the Within-Class Pupil Samples

As was described in the observation component section, a carefully defined sample was drawn for the purposes of identifying differences in instructional procedure which might correspond to resource allocation condition (i.e., concentrated vs. saturate "treatments"). The differences were sought in terms of pupil characteristics (gender, ethnicity, and relative disadvantage), which could relate to outcome patterns. Therefor the observation component was primarily designed to obtain evidence of differential effectiveness of either of the two resource targeting strategies in terms of individual differences among pupils.

The basic within-class samples were selected to maximize EDY differences, while retaining an appropriate gender and ethnic composition across classes. Specifically class rosters were prioritized in terms of pupil quartile on the previous year's MAT reading score. A sample of four pupils (two EDY, or Q1; two nonEDY or Q3)^{*} were drawn at random from each class so that within grade level the samples were reasonably well balanced on gender and ethnicity as well. (Two alternate pupils were also designated—one EDY and one nonEDY—within each class.) The resultant pupil samples constituted the targets for the individually focused instructional observation procedures, and for the pupil-focused analysis of effects.

Even with these over-sampling precautions, problems of attrition and incomplete teacher data reduced the original sample of 288 pupils to a final sample of 219 pupils (56 teachers X 4 pupils/teacher should have yielded 224 pupils). These resultant overall and within-grade pupil samples are displayed in Table <u>10</u> and Table <u>11</u> on page 31.

Because very few Alum Rock elementary pupils score in Q4, Q3 was selected as the more representative nonEDY population.

ERIC Pruil Text Provided By ERIC

45.

Multiple linear regressions were performed on samples within grade-level to identify reliable context and process covariates of pupil achievement. The available measures for each pupil included prescore (previous MAT standard scores for sub and total tests), design variables (EDY status, resource treatment, ethnicity, gender), and a set of process observation descriptors. Means and standard deviations on these variables are displayed in Appendix B.3. Consequently, within each grade level postscores were regressed on available process and context variables, including the corresponding prescore.

The identification of relevant process covariates of outcomes was enhanced by attaching differential weight factors to the process variables in the stepwise procedures (i.e., "process" variables were weighted more heavily than prescores such that the regression analysis was "forced" to consider process variables before stepping prescores into the equation).

A) Grade 2 Regression Results

Results of the regression analyses on the four outcome measures for the second grade sample are summarized in Appendix D.1. As demonstrated by these results, pupil outcomes apparently are not well explained by available observation measures. The highest proportion of outcome variance explained by prescores and process measures is for the Total Reading score; however, this only amounts to 53.7%. Even for those process measures which apparently account for significant proportions of criterion score variance (e.g., typical role-group leader and total number of minutes the pupil was observed to be idle), the anticipated relationships materialized differently than expected. For examp time idle positively relates to outcome score, indicating that pupils with higher observet idle time score higher on the post-tests. This probably indicates that teachers spent more time with EDY pupils, therefore neglecting nonEDY pupils, at least during the onetime pupil observation session.

B) Grade 4 Regression Results

Formally identical regression analyses were performed on fourth grade pupil sample outcomes (see Appendix D.2). Although the amount of outcome variance was only slightly greater for these 113 grade 4 pupils (maximum = 57.9% for Total Reading Score), the

significant process predictors are apparently in accord with findities reported in relited research. Specifically, there is consistently an inverse relationship between the amount of time these pupils were engaged in noninstructional activities and outcome scores (p<.05). In other words, the more observed noninstructional time, the lower the subsequent scores.

Other measures which reliably account for outcome-specific_results are both the relative amount of time pupils were observed to be in the "receiving directions or assignments" mode (again negatively relating to outcomes) and the overall instructional grouping (whole class vs. staggered). The data apparently indicates that whole-class instruction is more effective.

Regressions without EDY Status as a Context Variable

When EDY status is excluded from the set of available regressor (i.e., context and process) variables, the results of the regression change in terms of both the relevant process-outcome predictors and the magnitude of explained criterion variance. The results of the second grade regression under this constraint indicate that patterns of materials usage (both in terms of amount and variability) marginally effect outcomes and accounts for, at most, 6% of the criterion variance (see Appendix E.1). The instruction modes observed in use during the 30-minute pupil observation sessions were even less effective (accounting for generally not more than 3% of the outcome variance).

This pattern of results also occurred for regressions of fourth grade measures on process variables (excluding EDY status), (see Appendix E.2). Essentially, the only cle distinction between second and fourth grade regression results is the total outcome variance explained (maximum for grade 2 = 46%; maximum for grade 4 -= 68%), which is a direct consequence of the stronger pre-post correlations observed for grade 4 data. The process variables collectively never exceed 10% explanation of criterion variance, regardless of measure or grade level. Accordingly, these "best available" process covariates were included with the relevant prescores in the subsequent analysis of variance/covariance of pupil-level learning outcomes.

It should be noted that although EDY status was defined as a context variable, it actually strongly aliases prescores, which are used to establish EDY status.

47.

Analysis of Covariance Results

Results from regressions on each within-grade level sample were used to define the most relevant covariables for each outcome measure, which would be subsequently analyzed in terms of the sampling design. Moreover, two forms of the outcome measures were analyzed:

(1) the Spring 1978 MAT standard scores

(2) Spring 1977 to Spring 1978 MAT "change" scores (standard). Results are reported separately within each grade level sample.

A) Grade 2 Spring 1978 Outcomes

The results of the four-way analyses of covariance for the second grade pupil sample for the Word Knowledge, Word Analysis, Reading Comprehension, and Total Reading measures are presented in Appendices F.1 through F.4. Although some differences in significant main effects and interactions are obtained from measure to measure, the general pattern of findings appears to be as follows:

- (a) Evidence for overall superiority of concentration or saturation did not approach statistical significance.
- (b) Even after adjusting for prescores as covariables, differences in outcomes in terms of initial EDY status remained highly significant (p<.001).</p>
- (c) Evidence of differential effectiveness of treatment (concentration vs. saturation) by EDY condition did not approach significance.
- (d) Only for Word Knowledge scores were reliable patterns of differential effects of treatments in terms of ethnicity or gender within EDY status found to occur.

Table 18

Spring Test Scores--Grade 2

		Concentrated (52.7)						
WORD KNOWI	LEDGE	EDY (46.8)	NON-EDY	(59.3)			
	• • • • • • • • • • • • • • • • • • •	Gir1	Boy	• Girl	Boy			
	Spanish	47.4	46.2	51.7	61.4			
	Caucasian	-	49.0	62.8	67.0			
	Black	52.0	-		57.0			
•	Other	-	46.0	67.0	51.0			
	Total	42.4	45.4	58.0	60.4			

49.9 "	48.1	58.9	60.6	53
-	` -	_ ·		
50.5	41.0	54.5		۰ ۱
67.0	50.2	53.0	63.0	
48.3	49.7	62.7	58.7	
Girl	Boy	Girl	Boy	
EDY (9.0)	NON-ED	<u>(59.7)</u>	
<u>Sa</u>	iturate	<u>a (53.8)</u>		

÷

Concentrated (51.2)

	Concentrated_(51.2)					
EDY (44.8)	NON-ED	Y (58.3)			
Girl	Boy	Girl	Boy			
sh 44.8	43.2	57.4	53.4			
ian -	45.8	60.3	68.0			
51.3	-	-	53.0			
. –	31.0	04.0	62.0			
46.2	43.5	60.2	56.7			
	Girl sh 44.8 sian - 51.3 -	<u>EDY (44.8)</u> <u>Girl Boy</u> Sh 44.8 43.2 Sian - 45.8 51.3 - - 31.0	$ \underline{EDY (44.8)} NON-EDY \underline{Girl Boy Girl} \underline{Girl Boy Girl Boy Girl} Girl Boy Girl Bo$			

Sa	turate	<u>a (49.3)</u>		
EDY (4	3.6)	NON-ED	<u> (56.4)</u>	
Girl	Boy	Girl	Boy	
41.7	44.7	, 58.3	56.0	
50.0	48.2	54.3	58.4	
42.0	37.5	49.5	-	
-	-	-	- ,	
42.4	44.9	55,8	57.2	50.:

		Co	ncentrat	ed (50.8))	Saturated (49.7)				
TOTAL READI	EDY (44.9)		NON-EDY (57.4)		EDY (44,4)		NON-EDY (56.4)			
		Girl	Boy	Girl	Boy	Girl	Boy	Girl	Boy	•
	Spanish	44.9	43.9	58.8	53.8	42.7	44.6	58.8	55.8	
2	Caucasian	-	44.0	60.7	68.5	51.0	48.0	52.3	` 59.0	
· .	Black	51.7	-	– .	53.0	43.5	41.5	49.0	_	
•	Other	-	38.0	74.0	56.0	-	-	-	-	
	Total	46.4	43.5	58.9	56.1	43.4	45.4	55.6	57.4	50.

ERIC

To aid in interpreting these patterns of results, mean scores by design condition are summarized by each outcome measure and are presented in Table 18. Clearly, overall averages by treatment rarely differ by more than one standard score unit, whereas EDY/nonEDY differences are quite large (10 or more standard score points) and relatively consistent across treatment conditions.

In terms of within-district norms, these outcomes can be interpreted as follows:

- (a) Concentrated EDY pupils averaged at about the 36th percentile; concentrated nonEDY pupils at the°75th. Saturated EDY pupils averaged at about the 33rd percentile; saturated nonEDY at about the 73rd. At most, concentration has provided about 5 percentile points educational advantage to these second-grade pupils.
- (b) A weak differential trend suggests that saturated methods might be more beneficial to boys (59th percentile) than girls (52nd percentile), with the opposite being the case for concentrated methods (boys = 55th percentile; girls = 63rd percentile). However, these patterns failed to reach statistical significance.

Grade 2 Difference Score Analyses B)

Total

12.5

In terms of the net educational benefit produced by these "treatments," analysis of the patterns of available pre-to post-score change was conducted using relevant process variables as covariates (as identified through separate regression analyses conducted on pre-post change scores). The results of these analyses are presented in Appendices G.1 through G.3 with the corresponding difference mean scores summarized in Table 19.

Table 19

8.1

11.

9.

13.6

15.9

9.5

7.0

	N	P	re-Post	Change	Scores	-Grade 2		,	
		-	ncentra	ted (10.	8)		Satura	ated (11.	5)
WORD KNOWLE	DGE	EDY ((11.7)	NONEDY	(9.8)	EDY (1	4.7)	NONEDY	(7.5)
•	•	<u>Girl</u>	Boy	Girl	Boy	Girl	Boy	Girl	Boy
. •	Spanish	12.9	10.0	6.0	9.8	15.3	12.0	16.6	6.0
	Caucasian	<u>.</u>	12.2	12.7	19.0	24.0	17.6	-7.3	10.6
	Black	11.3		'	14.0	15.0	10.7	-5.0	
	Other	 ``	12.0	20.0	-3.5				

10.2

10.9

		Co	ncentra	ted (9.5	<u>) </u>	Saturated (9.2)				
WORD ANALY	<u>SIS</u>	EDY (8	.8)	NONEDY (10.2)		EDY (9.4)		NONEDY (9.0)		
		Girl	Воу	Girl	Воу	Girl	Boy	Girl	Boy	
	Spanish '	8.9	4.5	6.7	12.0	10.4	6.8	11.0	6.0	
	Caucasian		14.0	10.8	14.5	5.0	12.0	7.7	10.2	
· .	Elagk	10.6	00		10.0	13.0	8.7	11.5	<u> </u>	
۰ لو د	Other		8.0	13.0	6.0					
2 million	Total	9.3	8.5	8.9	11.4	10.2	8.7	10.2	7.7	

دي . 		Co	ncentra	ted (11.	<u>6)</u>	, 1—————	Satura	ited (10.	9)	_
TOTAL REAL	DING	ÈDY (1	2.8)	NONEDY	(10.5)	EDY (1	1.2)	NONEDY	(10.4)	.
. / .		Girl	Boy	Girl	Boy	Girl	Boy	Girl	Воу	•
	Spanish	13.4	11.0	9.2	7.7	10.3	9.7	14.3	10.8	
÷ .	Caucasian	:	13.2	12.7	22.5	11.0	16.4	0.7	12.4	
	Black	16.0	· `	、 	10.0	10.5	9.5	5.5		
c	Other	•	8.0	31.0	-2.5					
	Total	14.0	11.7	12.8	8.5	10.4	12.1	9.4	11.6	11.

Because of changes in subtest content across years, not all outcome measures. had corresponding prescores.

These results show that strong differential gains occurred only for the Word Analysis subtest (p<.01). These gains were most marked for EDY pupils in saturated class and least marked for nonEDY pupils in saturated classes (see Table 19). Based on local norms, the EDY pupils in saturated class apparently moved from the mean percentile rank of 30 in theSpring of 1977 to the m an percentile rank of 37 in the Spring of 1978 in terms of Word Knowledge skills. NonEDY pupils in saturated classes lowered their score from the 86th percentile in Spring 1977 to the 77th percentile in Spring 1978. The results for both EDY and nonEDY pupils in the concentrated classes on the same subtests are as follows:

	<u>Mean Per</u>	centile Rank	(Word Knowledge)
а. Р	Spring 1977	Spring 1978	Net Change
Concentrated EDY	33	35	+ 2
Concentrated nonEDY	80	77	- 3

A table of net percentile rank change (again based on within-district norms) on Total Reading scores for the second grade sample is as follows:

•	Mean Pe Spring 1977	rcentile Rank (7 Spring 1978	Total Reading) Net Change
Concentrated EDY	· 27	36	+ 9
Concentrated nonEDY	82	79	- 3
Saturated EDY	27	34	+ 7
Saturated nonEDY	85	75	-10

Therefore, using relative within-school district status as the effectiveness criterion concentration is apparently a superior treatment to saturation. This interpretation cannot be advanced unequivocably, however, since these score patterns are at least partly influenced by the regression-toward-the-mean phenomenon inherent in pre-post analyses.

52.

C) Results of the Grade 4 Analyses

Total

60.4

56.1

Analyses formally identical to those reported above for the second grade sample were conducted on the fourth grade sample (102 pupils); however, the Word Analysis Subscore was not available for the reading test used at grade 4 (the MAT elementary level), and the reading comprehension difference scores were unavailable for this sample. The results of analyses of Spring 1978 outcome measures are presented in Appendices H.1 through H.3. The corresponding averages are summarized in Table 20.

		Spring	1978 Test	Scores-	-Gr	ade 4		÷	
WORD KNOWLEDGE	Co	ncentra	<u>țed (66.6</u>)	•	\$	aturate	ed (64.3)	
WORD RIVOWLEDGE	EDY (60.2)	NONEDY (76.4)			EDY (56.6)		NONEDY	(72.7)
<u>_</u>	Girl	Boy	Girl	Boy	$\ $	<u>Girl</u>	Boy	Girl	Воу
Spanish	61.7	60.2	76.0	73.0		55.9	54.1	71.7	69.5
Caucasian	60.7	66.0	74.7	72.0		60.0	50.5	78.8	74.5
Black	55.5	56.5	73.0	101.5	.	55.5	59.0		73.0
Other	59.0 .		64.0	70.0			59.0		
Total	60.3	60.1	74.2	80.6		56.2	55.1	74.3	71.4
PEADING COMPENSION	Conc	entrate	ed (64.6)				Satura	ted (63.	3)
READING COMPREHENSION	EDY (S	58.3)	NONEDY	(73.8)		EDY (5		NONEDY	
	Girl	Boy	Girl	Boy		Girl	Boy	Girl	Boy
Spanish	61.2	53.4	75.8	73.7		48.7	56.5	73.0	70.4
Caucasian	.60.7	77.0	71.2	64.0	. 6	52.0	49.5	74.2	72.5
Black	52.0	60.5	74.0	89.5	5	53.0	55.0		74.0
Other	71.0		60.0	72.0			49.0		

Spring	1978	Test	ScoresGrade	4
0		1000	DCDIESDIAUE	-

Table 20

TOTAL READING		Co	ncentrat	ed (65.0)	<u>Saturated</u> (63,1)					
		EDY (58.2)		NONEDY (74.9)		EDY (53.1)		NONEDY (71.9			
•	· · · · · · · · · · · · · · · · · · ·	Girl	Воу	Girl	Boy	Girl	Воу	Girl	Воу		
	Spanish	60.3	56.5	76.0	73.0	51.6	53.8	71.9	69.1		
· · · · · · · · ·	Caucasian	58.8	69.0	72.2	67.0	59.0	49.0	76.0	72.8		
	Black	53.0	56.5	73.0	97.0	53.0	56.5		73.0		
	Other	62.0		61.0	71.0	-	54.5				
	Total	58.9	57.4	72.8	78.7	52.6	53.5	73.4	70.7	6	

72.3

76.6 50.9 53.8 73.4

65.5

63.8

71.5

53.

As distinct from the second-grade results, clear differences associated with treatment are found for these grade 4 pupils on both the Word Knowledge and Total Reading scales. Moreover, when the preceding scores for these pupils are used in analysis, as is done in the analyses of difference scores reported in Appendices I.1 and I.2 and summarized in Table 21, the effects become even more marked.

'Table 21

Pre-Post (Change) Scores--Grade 4

5.

	·····		nčentra	ted (6.2)	· ·	Saturated (4.1)				
WORD	KNOWLEDGE	EDY (6.0)		NONEDY (6.4)		EDY (5.6)		NONEDY (2.6)		
		Girl	Boy	Girl	·Boy	Girl	Воу	Girl	Boy	
	Spanish .	9.2	4.6	5.2	2.7	4.4	6.6	1.3	-2.0	
	Caucasian	7.8	9.0	4.8	5.0	8.0	2.5	8.0	6.5	
	Black	0.5	5.0	-4.0	34.5	4.0,-	8.5		7.0	
	Other	2.0		-6.0	6.0		7.5			
	Total	7.0	5.0	3.1	12.6	4.6	6.4	3	1.7	

	17 - E	Cor	ncentrat	ed (5.8)			Satura	ated (4.1)
TOTAL READING		EDY (6.6)		NONEDY (4.7)		EDY (5.0)		NONEDY (3.2)	
	•	Girl	Воу	Girl	Boy	Girl	Boy	Girl	Boy
	Spanish	10.3	3.7	4.8	0.0	3.0	8.0	1.7	0.9
•	Caucasian	9.5	12.0	2.3	1.0	6.0	3.5	5.2	6.0
	Black	-1.0	8.5	-8.0	27.5	2.0	6.0		8.0
	Other	6.0		-1.0	8.0		6.5		·
	 Total	/ 8.2	5.0	2.2	9.1	3.1	6.7	3.0	3.4

The concentrated treatment provided more desirable effects than the saturated treatment. These patterns of outcomes can be displayed most clearly as relative effect shifts in within-district percentile rankings. Specifically, for Total Reading measure the pattern is as follows:

· · · · · · · · · · · · · · · · · · ·	Mean Perc	entile Rank (Tota	al Reading)
- -	Spring 1977	Spring 1978	Net Change
Concentrated EDY	18	30	.+ 12
Concentrated nonEDY	91	88	- 3
Saturated EDY	18	19	+ 1
Saturated nonEDY	90	81	ʻ

ERIC PENILEME PROVIDENCE ERICE

54.

 $5\overline{s}$

Essentially, the relative within-district rank of EDY fourth-grade pupils receiving concentrated services increased by an average of 12 percentile rank units, whereas their nonEDY counterparts stayed relatively stable (declined 3 percentile rank units). In comparison, the EDY fourth-grade pupils receiving saturated services increased their ranking only 1 percentile rank unit over their relative within-district ranking the preceding year, while their nonEDY counterparts declined an average of 9 percentile rank units.

A similar analysis of relative shifts on the Word Knowledge scale reveals the following patterns:

	Mean Percer	ntile_Rank (Word	Knowledge)
	Spring 1977	Spring 1978	Net Change
Concentrated EDY	30	34	+ 4
Concentrated nonEDY	88	88	0
Saturated EDY	20	17	- 3
Saturated nonEDY	88	78	- 10

Again there is an upward shift for concentrated EDY, a downward shift for saturated EDY, virtually no shift for concentrated nonEDY, and a marked decline for saturated nonEDY.

One possible explanation is that the teachers in the saturated classes frequently interpreted their assignment as providing resources equally and uniformly to all pupils, rather than making resources available to all pupils on the basis of diagnosed need. Therefore, teachers in the saturated fourth grade classes mechanically implemented the compensatory services (the data indicates the teachers in the saturated classes did this significantly more frequently than teachers in the concentrated classes). Consequently, neither EDY or nonEDY pupils benefited.

5.)

55.

ERIC

Alternatively, the data indicates that the teachers in the concentrated classes provided differential services according to pupil need; therefore, both EDY and nonEDY pupils benefited.

An alternative interpretation of these results, also indicated by the observation findings, is that concentrated teachers were more likely to segment their class, and make use of aides and resource centers to intensify services directed at the poor-performing pupils. Therefore, these findings might indicate that not only did the EDY pupils receive appropriate individualized assistance but also the teacher appropriately differentiated methods and materials to the nonEDY pupils. Since many teachers in the fourth grade saturated classes apparently felt they were expected to treat all pupils equally (even though they clearly could not), it is reasonable to conclude that they were less inclined to individualize, by using either pullout/resource center facilities, differential instructional methods or materials matched to pupil ability. If this second interpretation is correct, then the effect of asking teachers to "saturate" services may, in their perception, be equivalent to asking them to homogenize instruction.

Ga

IV. Multiple Regression Analyses Results

Even though the analyses of covariance indicate a substantial benefit may be associated with concentrating compensatory services, particularly in the upper elementary grades (grade 4), such analyses do not identify the instructional componentsassociated with these benefits. In addition, these analyses do not evaluate other features of instructional programs (e.g., use of human and material resources, engaged time patterns, teacher style, etc.) which may account for additional differences in reading achievement. To facilitate additional analyses and evaluation of these features, a series of stepwise multiple linear regressions were performed on outcome measures within each grade-level sample. Basically, this analysis attempts to discover: What are the process and context characteristics at the classroom level which best account for differences in mean achievement on criterion tests? The results of these analyses, confirm and extend the results of the covariance analyses.

Grade 4 Class-level Regressions

Outcome measures (Spring 1978 Total Reading and subtest average standard scores) for the 28 fourth-grade classes having complete data (interview, observation, and test data) were separately regressed on process and context indicators. A forward stepwise procedure was used which restricts inclusion to significant predictor variables. However, once a variable was included, it remained in the equation regardless of subsequent changes in its predictive significance.

Three criterion tests available for the grade 4 sample (Total Reading, Word Knowledge, and Reading Comprehension) are reported.

57.

Total Reading Achievement

Average standard scores on the Total Reading scale for the 28 fourth grade classes yielded an unusually high degree of statistical explanation when regressed on context and process measures. Altogether, 13 measures entered into the explanation of Reading score outcomes, although two eventually dropped below the criterion significance level (.05). The resultant multiple correlation is .989, indicating that nearly all (i.e., 97.7%) of the variability in outcome measures has been "explained" by these relevant context and process measures. Even adjusting the solution for the number of predictors, the explanation still accounts for better than 95% (adjusted R^2 =.956) of test score variance.

Although the regression solution has identified a linear combination of both context and process determinants, for interpretive convenience they are presented separately. Two statistics are reported for each determinant: a standardized <u>Beta</u> value, indicating the relative potency of the determinant (i.e., the expected unit change in the criterion for a unit change in the determinant, all other determinants held constant), and a step (or univariate) F value, indexing the reliability of the determinant (F>4.5, p<.05).

Context Determinants

Fourth grade classes generally achieved higher overall reading scores to the extent they consisted of:

		Beta	Stepwise <u> </u>
(a)	proportionately fewer Spanish surname pupils	269	11.20
(b)	proportionately more Anglo/caucasian pupils	.242	9.14
(c)	proportionately fewer girls	179	12.69

Conditions which approached significance in relating to above-average Total Reading achievement were:

•	s by	Beta	Stepwise G
•	the use of whole class (as opposed to sta reading instruction	aggered) 116	3.94
•	a higher average age of the class	.021	1.03

Process Determinants

In addition to these context determinants, the following instructional characteristics added significant explanation of Total Reading achievement (i.e., represent the significant process determinants):

•		Beta	Stepwise F /	
· (a)	the classroom received concentrated services		33.44	
(Ъ)	individual pupils were observed to make a wider use of materials	.685	175.10	
(c)	teachers were observed to make proportionately greater use of EDY materials	•527	76.16	q
(d)	more observed variablity (viz individualization in the amount of time pupils were being managed or receiving directions	.313	54.47	••
(e)	pupils were <u>less often</u> observed making use of EDY materials	303	28,79	
(f)	teachers reflect d a more thorough understanding and correct implementation of their respective resource strategy (concentrated or saturated)	.348	20.29	•
(g)	teachers were observed to be more variable in the amount of time spent in directive roles	.233	13,45	
(h)	teachers were less extensive in their affective responsiveness to pupils	153	6.07	•

This analysis indicates that the class achieving the highest performance level has relatively more older caucasian males, relatively few young Spanish surname females, and is led by a teacher who concentrates resources on EDY pupils, individualizes instruction and the assignment of materials, more clearly understands resource management and the concentration/saturation experiment, balances activities between giving direct instruction to small groups vs. providing directions to pupils for self-instruction, and uses positive/negative feedback more conservatively (or selectively).

Word Knowledge

Analysis of mean fourth grade classroom standard scores on the Word Knowledge subscale yielded a high explanation which was strikingly similar to that reported for the Total Reading scores. Specifically, the overall multiple R on eight regressor variables was computed as .956, showing better than 91% of criterion variance had been "explained" by these eight predictors (R^2 =.914). The adjusted R^2 for this solution is .877, and the F value for this equation is 25.10 (df=i,19). Again, the interpretation of this result is presented in terms of context and process variables separately.

Context Determinants

In terms of context variables, mean performance on Word Knowledge subtests Stepwise Beta F

(a)	the classroom was made up of older pupils	. 190	4.05
(b)	the classroom consisted of an above-average propertion of caucasian/anglo pupils	.331	7.70

In addition, context variables which, originally significant, generally related to score patterns (but which are dominated by process variables) are: Stepwise

		<u>Beta</u>	<u>r</u>	
•	lower proportion of Spanish surname pupils	190	2.75	
٠	whole class (as opposed to staggered) reading instruction	151	3.90	

Process Determinants

The cumulative explanation available through these context determinants is, at maximum, 42%. Nearly 50% additional explanation is found with the four significant process determinants, which are interpreted as indicating that classroom score above average to the extent: $\frac{F}{Beta} = \frac{F}{F}$

			the second s
(a)	the teacher concentre 1 resources	.314	18.47
(Ъ)	pupils were observed to use a larger number of materials over the course of instruction	.586	56.78
(c)	teachers were observed to make use of a higher proportion of EDY materials	.296	15.48
(d)	more variability was observed in the amount of time teachers spent providing pupils with directions	.331	22.49

0.

Reading Comprehension

Average standard scores on the MAT Reading comprehension subscale were also regressed on context and process variables. As with the Word Knowledge scores, results for this analysis are quite similar to the total score results at a general level, with only minor variations in specific determinants. An overall solution involved nine variables, with a multiple correlation of .956, accounting for 91.5% of the total variance on Reading comprehension scores. The adjusted R^2 is .872, with an \tilde{F} on regression of 21.46 (df=9,18).

Context Determinants

The interpretation of this result in terms of context determinants is that high Reading comprehension scores were obtained for classrooms consisting of:

		Stepwise
(a) more pupils of above-progress and	Beta	F
peptie of above-average age	.240	10.25
anglo/caucasian pupils	.340	7.66
(c) fewer pupils of Spanish surname*	119	0.94

Collectively, these context determinants account for a maximum of 39% of all criterion variance.

Process Determinants

The remaining 52% of criterion variance is explained by the six process determinants, which indicate that mean scores on Reading comprehension increase to the extent:

		<u>Beta</u>	Stepwise F
(a)	teachers concentrate services	.372	17.70
(Ъ)	pupils were observed to make use of a larger number (variety) of materials	.793	-
(c)	greater variation in group leadership was observed for individual pupils (e.g., use of aides, peers, etc.)		74.51
(d)	teachers were observed to make greater overall use of EDY materials	,229	5.50
(e)	more variation existed in time spent giving individual pupils directions	.216	9.02
(f)	less variation over time was found in the rela- tive amount of pupil time spent in seatwork and self-instruction	• 4 1 0	8.22
		204	3.79

Originally significant, but subsequently aliased by percent caucasian.

61. ·

Grade 2 Classroom-level Regressions

Findings from regressions of criterion scores on context and process measures for second grade classes demonstrate neither the regularity nor the strength of association that was indicated by the findings from fourth grade classes. Again using classroom aggregated statistics on each category of measures (context, process, outcomes), separate stepwise regressions were performed on the Word Knowledge, Reading Comprehensio and Total Reading means for the 28 second grade classrooms. The results of these analyse are described and interpreted as follows:

Word Knowledge

The analysis of context and process determinants of Word Knowledge produced a surprising result. Approximately 74% of the criterion variance $(R^2=.745)$ is explained by four variables (multiple R=.863, F=16.77, df=4,23) and can be interpreted as showing second grade classrooms averages on Word Knowledge subtests increase to the extent:

-		<u>Beta</u>	Stepwise F
(a)	the classroom was composed of a smaller porportion of black pupils	430	13.91
(ь)	the teacher was observed to make below-average use of EDY materials relative to all materials	561	23.64
(c)	the teacher was observed to be more variable in the amount of time spent giving directions	.219	4.24
(d)	the teacher's previous class was above average on Word Knowledge	.333	9.81

One reasonable interpretation of this pattern of findings is that, for these second grade classrooms, Word Knowledge achievement is more strongly determined by characteristics of the pupils than by instructional methods <u>per se</u>, and furthermore, that teachers are adapting their methods and materials to these contextual differences (i.e., classroom composition). Since prior-year class score also reliably accounts for current outcomes, this possibly is evidence of a teacher effect (i.e., certain teachers are consistently associated with high-achieving classes, others with lowachieving classes). However, a more feasible explanation is that this effect reflects stable population differences associated with school attendance areas within the

district. These attendance area population differences would be manifest as differences in relative EDY composition at the classroom level, and would necessarily show up as constant differences in classroom achievement. Furthermore, this interpretation is consistent with EDY-use findings: namely, classes consisting of low percentages of EDY pupils would be expected to make less frequent use of EDY materials, and vice versa.

This regression result indicates, that, in the second grade, the use of instructional procedures (methods and materials) apparently do <u>not</u> overcome learning⁶ differences associated with socio-cultural group membership (perhaps as aliased by school attendance areas), at least with respect to Word Knowledge achievement measures. It does suggest that teachers are targeting resources (EDY materials) to perceived pupil needs; however, this targeting is highly correlated to ethnic group membership (again a correlate of school attendance areas).

Reading Comprehension

Results for regression of second grade Reading Comprehension means on context and process variables essentially replicate those found with Word Knowledge, except that far less criterion variance is explained. Only two reliable "predictors" of second grade reading comprehension were found, accounting for less than 30% of the outcome variance (R=.546, R^2 =.298, F=5.31, df=2,25). They are:

	Beta	Stepwise <u> </u>
(a) the relative use of EDY materials by the teacher	479	7.96
(b) the mean reading comprehension scores obtained by the previous year's class	.109	4.22

Again, high average scores occurred for classes in which teachers make less use of EDY mateials and for whom the teacher's prior year's class also scored above average. This is consistent with the interpretation that context (or school) effects dominate the outcomes, event though the teachers properly target resources.

Note: We also considered the alternative interpretation that the use of EDY materials and resources serves to depress scores. But in the absence of a difference due to concentration vs. saturation of EDY materials, this interpretation is considered less tenable. Rather, as is suggested in findings for Word Analysis outcomes, it appears that a large proportion of second grade teachers simply refused to implement their prescribed treatment.

67

Word Analysis

The MAT second grade battery (Primary II) provides for an additional skill area described as Word Analysis. Regression analysis of this measure on process and context variables produced some surprising results.

Virtually all of the criterion variance on this measure has been "explained" by a combination of ten context/process descriptors (eight of which remain highly reliable). The multiple correlation is .981, accounting for over 96% (R^2 =.962) of the variation in Word Analysis mean scores (F=38.36, df=10,15). Even adjusting for the number of predictors, the explanation is still extraordinarily high (adjusted R^2 =.937). This result indicates mean scores on the Word Analysis subtest increase to the extent:

			<u>Beta</u>	Stepwise . F	\setminus
7	(a)	the class consists of a lower percentage of black pupils	487	61.45	•
	(Ъ)	the teacher uses staggered (as opposed to whole class) reading instruction	.040	0.31	•
	(c)	the teacher was observed to use proportionately fewer EDY materials	- 970	118.62	3
•	(d)	the teacher was more variable in the assignment of pupil seatwork/self-instruction	.433	59.73	
¥	(e) '	the teacher was <u>less</u> in compliance with her/his respective treatment condition	082	2.00	
	(f)	the teacher's prior class scored above average on the Word Analysis subtest	. 908`	106.60	
	(g) .	the teacher perceived her/his resource targeting guidelines (i.e., treatment) to be at variance with the district policy	.402	44.54	
- i	(h)	the proportion of girls in the class was above the overall sample average (.545	44.55	
	(i)	the teacher saw him/herself as primarily responsible for learning outcomes	.403	25.26	
((<u>†</u>)	the teacher tended to be more demonstrative in the use of positive and negative affective responses	.258	15.41	•

Basically, this result both reinforces and elucidates previous findings regarding determinants of Word Knowledge and Reading Comprehension scores in these second grade classes. Specifically, the context determinants account for better than 46% of outcome variance, with about 50% attributable to process characteristics (including teacher attitudes about the validity of the experiment). This corresponds closely with findings

64. .

 $6\overline{a}$

for the fourth grade sample, where process measures generally account for about 50% of the criterion variance. This result is consistent with the interpretation that school/ community characteristics significantly influences teacher strategies and subsequent outcomes in this second grade sample.

Even more startling, however, are the findings relating teacher attitudes and behaviors (regarding the targeting of resources) to subsequent outcome patterns. The teachers with higher attaining classes apparently thought that the "treatment" definitions were vague and arbitrary; therefore, their behavior was influenced by this opinion. Moreover, these teachers attribute the achievement of their pupils to their own teaching methods (rather than to resources and administrative support).

These second grade teachers apparently assert that they know how to optimally allocate resources, and to a considerable extent the findings corroborate this assertion The strong negative influence of ethnic composition of the classroom to outcomes is still troubling; clearly, the relationship between ethnicity and learning has not been overcome by these teachers, and needs further investigation.

Total Reading Score

The overall relationship between Total Reading score and context/process character istics is shallow for these second grade classrooms, and is far more difficult to interpret than the subscore findings. Only two variables were found to reliably account for Total Reading scores at the second grade:

	Γ	Beta	Stepwise F
(a) observed use of EDY materials		473	· 8.20
(b) mean age of pupils	all is	314	- 4.62
Approximately 32% of outcome variance is account	nied for by thes	e measures	$(R=.564, R^2=.318)$
F=5.83, df=2,25), and as the beta coefficients	show, in both c	ases the p	redictor rela-
tionships are inverse (lower mean age and use o	of EDY materials	account fo	or higher
outcome scores).		•	•

65.

One plausible interpretation is that this finding reflects, in part, the confounding of second grade retention policies. In other words, it is conceivable that classes of above-average pupil mean age contain disproportionate numbers of slow learners who are repeating the second grade. This, in turn, would account for the negative relationship between mean pupil age and mean achievement. Unfortunately, at the time of this writing, data are not available to corroborate this interpretation.

15

SUMMARY CONCLUSIONS

Alum Rock Union Elementary School District participated as one of 11 national demonstration sites to assess the relative educational effects of variations in school-wide targeting of compensatory services. The effects of the concentration and saturation methods provide the general focus of this report.

The data used in assessing the relative effects of these two resource provision conditions were provided by establishing a matched sample of 18 schools which were randomly assigned to "saturation" or "concentration" of EDY resources. Two basic questions guided the analysis of the data:

- Does saturation or concentration of compensatory resources and services--to the extent such occurred in this study-related to reliable and meaningful differences in basic reading skill attainment?
- 2. What are the contextual and procedural (instructional) conditions which account for differences in reading skill attainment?

An additional objective was to determine the central and peripheral effects of the allocations of these treatments on classroom practices and pupil learning (reading, as measured by MAT). The objective was to determine the characteristics of pupils, resources, and instructional procedures which combined to account for learning outcomes (i.e., reading skills, as measured by the MAT).

Some of the major findings of the observational component were:

1. Teacher Roles

More than three roles were observed in only 10% of the teacher-focused observations. The maximum number observed was five. Diversity of teacher roles did not differ reliably across treatment group or grade levels. The most common teacher activities observed were oral or silent reading and reviewing (24%), drill (23%), classroom management (15%), and assigning tasks (9%). Together these four activities represented 71% of all teacher roles observed. These roles were predominant across treatment groups and grade levels.

2. Pupil Roles

Many of the pupil roles were the counterpart to teacher roles. The most common activities involving pupils working with the teacher were oral/silent reading and review (22%) and drill (20%). Receiving assignments and participation in classroom management activities represented 8% and 9% of all pupil roles, respectively. Two additional roles were fairly common among pupils: seatwork (completing assignments-ll%); and transition (waiting for a new task or the teacher's attention--6%). Together these activities accounted for three-fourths of all the pupil roles observed.

3. Engaged Time

A measure of engaged time was generated for each of the 219 pupils by calculating the total number of minutes each spent in all roles classified as instructional. Second and fourth-grade pupils spent an average of 87% and 90% of their time, respectively, engaged in activities directly related to instruction. Reliable differences between treatment groups were not found at either grade level. EDY and non-EDY students in both saturated and concentrated classes generally had a high proportion of engaged time. In the fourth-grade concentrated classes, EDY pupils spent an average of 95% of their time in instructional roles, while the non-EDY pupils spent an average of 85%.

4. <u>Teacher-Pupil Interaction</u>

The teacher-initiated interactive mode clearly dominates, accounting for an average of 70% of teacher time. The directive and facilitative modes together represent an average of 25% of teacher time. Discussion and social interaction are comparatively rare. Pupils working with the teacher during reading instruction spent an average of 65% of their time in the responding mode. Most of their remaining time was spent completing seatwork assignments or working on self-instructional activities. A comparison of the teacher and pupil profiles showed strong similarities in the relative proportion of time spent in corresponding teacher-pupil modes. Teachers in the saturated classes spent 70% of their time in

72

the interactive mode. Correspondingly, the pupils in those classes spent 72% of their time in the interactive mode.

Teachers in concentrated classes spent more time monitoring or assisting their pupils than teachers in the saturated classes. Pupils in concentrated classes spent an average of 28% of their time in seatwork and self-instruction. This represents almost twice the percentage spent by pupils in the saturated classes. Teachers in concentrated classes spent a significantly greater amount of time facilitating activities than teachers in saturated classes.

One possible explanation is that teachers in concentrated classes used various modes of instruction as a technique for concentrating services on EDY pupils. Unfortunately, data from the individual observation instrument did not show systematic differences in use of time by EDY and non-EDY pupils in the concentrated classes.

5. Materials

The average number of different materials used during a ten-minute epdisode was 2.00. An average of 2.20 materials per episode were used in concentrated classes, which is significantly higher than the average of 1.86 materials used in saturated classes. Teachers in concentrated classes were more likely to use materials purchased with EDY funds.

Diverse use of materials apparently did not differ as a function of grade level, treatment condition or EDY classification. However, fourth grade pupils in concentrated classes used EDY materials more often than fourth grade pupils in saturated classes.

6. <u>Teachers' Interpersonal Style</u>

Teachers' interpersonal style was also observed during the teacherfocused observations. Behavioral data for each of the 56 teachers were combined across observations to form several indices of the teachers' interpersonal style or responsiveness.

73

69.

Even though virtually all teachers displayed some supportive affect, the display was relatively infrequent and at a low level of intensity. Most teachers took care to praise the students only when appropriate. The vast majority of teachers rarely commented or acted in a manner which indicated disapproval of pupils' work or behavior.

Correlative analysis of the affect variables revealed general findings regarding teachers' interpersonal style: Teacher praise and approval are not related to criticism and disapproval. Essentially, teachers who scored relatively high on the positive measures were neither more nor less likely than other teachers in the sample to score high on the negative (nonsupportive) measures. Supportive and nonsupportive responsiveness apparently functions as relatively independent components of these teachers' interpersonal styles. With the sample of second and fourth grade teachers, the interpersonal style was found to be related more to grade than to treatment group, with teachers responding more frequently to second grade pupils.

7. <u>Instructional Modes</u>

Second grade teachers in concentrated classes apparently made some distinctions between EDY and non-EDY pupils. EDY pupils spent somewhat more time than non-EDY pupils in the responding mode. Non-EDY pupils spent more time receiving directions and completing seatwork and self-instruction tasks. Within saturated classes, the majority of time was fairly evenly divided between the responding mode and the seatwork/self-instruction mode for both EDY and non-EDY pupils.

The results from the fourth grade differed. Fourth grade teachers in both treatment groups did not make systematically different use of the three instructional modes for EDY and non-EDY. Within the fourth grade, patterns were very similar within treatment type but very different <u>between</u> treatment type. Pupils in concentrated classes spent an average of half their time in seatwork and selfinstruction. In saturated classes, teachers generally used the interactive mode for both EDY and non-EDY children.

74 .

70.

8. Group Size

Average group size was about 10 in both the second and fourth-grade classes. Group size did not differ reliably between treatment groups in either grade. In the concentrated classes, groupings do not appear to differ between EDY and non-EDY pupils. In saturated second-grade classes, however, non-EDY pupils generally worked in smaller groups than EDY pupils. This distinction was not found in the fourth grade saturated classes.

Results from the analysis of the outcome measures differ somewhat for the grades analyzed (2 and 4). More specifically, the results indicate the following:

1. Fourth Grade Results

A. Results of the analysis of covariance on the 28 fourth-grade classes point to the following conclusions:

For both the Total Reading and the Word Knowledge measures, concentrated services produced reliably greater mean scores than saturated services. These effects were evident before and after ajustment for process-context covariates (which include pre-scores). As distinct from second grade results, clear differences associated with treatment are found for these fourth-grade pupils on both Work Knowledge and Total Reading scales.

Mean differences on Reading scores (a comprehension sub test) favored the concentrated condition.

B. Fourth-grade multiple regression results point to the following conclusions:

The significant process predictors are apparently in accord with findings reported in related research. Specifically, there is consistently an inverse relationship between the amount of time these pupils were engaged in noninstructional activities and outcome scores (p < .05). In other words, the more observed noninstructional time, the lower the subsequent scores.

75

7].

Essentially, the relative within-district rank of EDY fourthgrade pupils receiving concentrated services increased by an average of 12 percentile rank units, wehreas their non-EDY counterparts stayed relatively stable (declined 3 percentile rank units). In comparison, the EDY fourth-grade pupils receiving saturated services increased their ranking only one percentile rank unit over their relative within-district ranking the preceding yéar, while their non-EDY counterparts declined an average of nine percentile rank units.

2. <u>Second Grade Results</u>

A. The covariate analysis for the second grade indicates the following:

Evidence for overall superiority for concentration or saturation did not approach statistical sign#ficance. Even after adjusting for pre-test scores as covariables, differences and outcomes in terms of initial EDY status, remain highly significant.

Evidence of differential effectiveness of treatment by EDY condition did not approach significance.

B. Results of the regression analysis for second grade outcomes point to the following:

For those process measures which apparently account for significant proportions of criterion score variance (e.g., typical rolegroup leader and total number of minutes the pupil was observed to be idle), the anticipated relationships materialized differently than expected. For example, time-idle positively relates to outcome score, indicating that pupils with higher observed idle time score higher on the post-tests. This probably indicates that teachers spent more time with EDY pupils thereby neglecting non-EDY pupils at least during the one-time pupil observation session.

76

Results of the analysis indicate a substantial benefit may be associated with concentrating compensatory services, particularly in the upper elementary

72.

grades (grade 4). Such analyses, however, do not identify the instructional components associated with these benefits. In addition, these analyses do not evaluate other features of the instructional programs which may account for additional differences in Reading achievement.

Using relative-within-school-district status as the effectiveness criterion, the concentration treatment is apparently superior to saturation. However, this interpretation cannot be advanced unequivocally since these patterns are at least partially influenced by regression toward the mean phenomenon inherent in pre-post analyses.

73.

APPENDIX A: DEFINITION OF ROLE TYPES

 $7\hat{\mathfrak{o}}$

74.

A.1 TEACHERS

A.2 PUPILS

A.3 FREQUENCY/TEACHERS

A.4 FREQUENCY/PUPILS

Eł

APPENDICES

APPENDIX A: DEFINITION OF ROLE TYPES

- A.1 Teachers
- A.2 Pupils
- A.3 Frequency/Teachers
- A.4 Frequency/Pupils

APPENDIX B: ANALYSIS VARIABLES

- B.1 Classroom-Level Variables
- B.2 Legend for Classroom-Level Variables
- B.3 Pupil-Level Variables
- B.4 Legend for Pupil-Level Variables

APPENDIX C: INTERCORRELATION MATRIX FOR CLASSROOM-LEVEL ANALYSIS VARIABLES

APPENDIX D: SUMMARY OF REGRESSION ANALYSIS D.1 On Second Grade Pupils

D.2 On Fourth Grade Pupils

APPENDIX E: REGRESSION RESULTS WITHOUT EDY STATUS

- E.1 Second Grade
- E.2 Fourth Grade

APPENDIX F: FOUR-WAY ANALYSIS OF COVARIANCE FOR SECOND GRADE PUPILS

- F.1 Word Knowledge - 1978
- F.2 Word Analysis 1978 F.3 Reading Comprehension 1978

- 1978

F.4 Total Reading

APPENDIX G: DIFFERENCE SCORE ANALYSIS FOR SECOND GRADE PUPILS

Ser.

- G.1 Word Knowledge
- G.2 Word Analysis
- G.3 Total Reading

APPENDIX H: FOUR-WAY ANALYSIS OF COVARIANCE FOR FOURTH GRADE PUPILS

- H.1 Word Knowledge - 1978
- H.2 Reading Comprehension 1978
- H.3 Total Reading - 1978

-79

75.

APPENDIX I: DIFFERENCE SCORE ANALYSIS FOR FOURTH GRADE PUPILS

- I.1 Word Knowledge
- I.2 Total Reading

Appendix A.1

DEFINITION OF ROLE TYPES--TEACHERS

Instructional

- 01 Assigning task 03 Discussing 04 Drilling 05 Facilitating AV 06 Facilitating manipulatives, games. 07 Facilitating oral/silent reading 08 Facilitating oral/silent reading, review 09 Facilitating student work 10 Instructing 12 Facilitating creative work 13 Reviewing 15 Testing assessing
- 16 Tutoring
- 23 Facilitating reading, writing
- 26 Praising student work
- 27 Facilitating other than reading

Directive

- 01 Assigning task
- 02 Disciplining
- 10 Instructing
- 14 Story telling, reading aloud
- 19 Managing
- 24 Reciting poetry25 Interrupted by office
- 26 Praising student work
- 28 Talking with parent

Teacher-initiated interactive

- 04 Drilling
- 07 Facilitating oral/silent reading
- 08 Facilitating oral/silent reading, review
- 13 Reviewing
- 15 Testing, assessing
- 16 Tutoring

Noninstructional

- 02 Disciplining
- 11 Interacting socially
- 14 Story telling, reading aloud
- 19 Managing
- 22 Doing Nothing
- 24 Reciting poetry
- 25 Interrupted by office
- 28 Talking with parent

Unable to classify

- 17 Can't tell, no English
- 18 Can't tell
- 21 Other, unclassified
- 20 No adult
- 00 Not applicable

Discussion and social interaction

- 03 Discussing
- 11 Social interaction

Assisting and monitoring

- 05 "Facilitating AV
- 06 Facilitating manipulatives, games
- 09 Facilitating student work
- 12. Facilitating creative work
- 23 Facilitating reading, writing
- 27 Facilitating other than reading

Teacher idle

22 Doing nothing

Unable to classify

- 17 Can't tell, not English
- 18 Can't tell
- 21 Other, unclassified

20 No adult

00 Not applicable

FRIC

Appendix A.2

DEFINITION OF ROLE TYPES--PUPILS

Instructional

01	Being assigned task 💦 🔿
03	Participating in discussion
04	Responding to drill
05	Using AV
06	Using manipulatives, games
07	Oral/silent reading
08	Oral/silent reading, review
09	Quiet task
10	Being instructed
12	Creative work
13	Responding to review
15	Being tested, assessed
16	Being tutored
21	Tutoring, work with peers
.24	Reading, writing
26	Other than reading
28	Reciting poems
29	Play rehearsal
31	Being praised, rewarded
32	Leave room, to resource center

Seatwork and self-instruction

- 03 Participating in discussion
- 05 Using AV 📃 🔪
- 06 Using manipulatives, games
- 09 Quiet task
- 11 Interacting socially
- 12 Creative work
- 21 Tutoring, working with peers
- 24 Reading, writing
- 26 Other than reading
- 27 Clean-up
- 28 Reciting poems
- 29 Play rehearsal (
- 32 Leave room, to resource center

Responding to teacher/group leader

04	Responding to drill
07	Oral/silent reading /
80	Oral/silent reading, review
13	Responding to review
15	Being tested, assessed
16	Being tutored

Noninstructional

- 02 Being disciplined
- 11 Interacting socially
 - 14 Listening to story
 - 19 Being managed
 - 20 In transition
 - 22 Not attending to task
 - 25 Leave room, personal reasons
 - 27 Clean up

Unable to classify

- 17 Can't tell, not English
- 18 Can't tell
- 23 Other, unclassified
- 00 Not applicable

Receiving directions

- 01 Being assigned task
- O2 Being disciplined
- 10 Being instructed
- 14 Listening to story
- 19 Being managed
- 31 Being praised, rewarded

Idle

- 20 In transition
- 22 Not attending to task

Unable to classify

- 17 Can't tell, not English
- 18 Can't tell
- 23 Other, unclassified
- 25' Leave room, personal reasons
- 00 Not applicable

ERIC Full Text Provided by ERIC

77.

- 81

APPENDIX B: ANALYSIS VARIABLES

- B.1 CLASSROOM-LEVEL VARIABLES
- B.2 LEGEND FOR CLASSROOM-LEVEL VARIABLES
- B.3 PUPIL-LEVEL VARIABLES

٩.

B.4 LEGEND FOR PUPIL-LEVEL VARIABLES

B.1 Classroom-Level Analysis Variables

84

^ .		~	
- 127	ade		
	aue	~	

٤

		•	Grade 2	т. К			Grade 4			M 1	
N					•	•	oraue 4			Total	
	VITIASLE	• MEAN	STANDARD DEV	CASES		MEAN	STANDARD DEV	CASES	MEAN	STANDARD DEV	CASES
	V17237	1.4643	0.5379	23		1.5000	0.5072	••			
) P372H	0.5571	0.1553	29		0.6010	0.14:5	28	1.4821	0,5042	55
	FILLOR	0.1120	0.0353	23		0.1224	0.0747	23	0.5540	0.1523	55 1
	REAUC	0.2525	0.1555	25		0.2122	0.1422	29	0.1172	0.0204	55
	GEN2	1.5103	0.1253	23		1.49:3		23	0.2324	0.1400	55
	2332	105.3563	17.7125	23		4.2619	0.0545	28	1.5017	0.0375/	55
	STAG	1.7143	0,45:3	23		1.6071		23	115.5541	19.7522	56 -
	xx77	50.4054	7.1354	23		4.0131	0.4973	23	1.6607	0.4778	55
	11.77	46.5732	7.3124	27	6	5.6370	7.1532	28	57.2092	9.8317	55
	5377	47.7182	7.2511	23		3.5334	7.1439	10 - •	47.0255	8.2532	37
	T01277 ×	47.5124	6.9717	23	0	2.8441 /	7.8451	23	55.6518	10.9533	55.
	TC5V01	1.7500	1.0753	23		2.0714	7.4922	23	55.2283	10.5105	55
	CTISV37	7.8727	1.5352	₁, 23 ^{°°}			1.1524	23	1.9107	1_1164	59
	TISV23	2.0357	1.5212	\$ 23		8.3729	1.1221	28	8.1429	1.4315	55
	TISVIS	11.0357	5.1457	23		1.8371	0.9315	23	T.5464	1.3131	55
	P35V01	1.7500	1.0753	23		9.8214	5.0410	23	10.4285	5.0340	56
	C713737	7.8729	1.6252	23		2.0714	1.1524	28	1.9107	1.1164	55
	PISV03	2.0357	1.6212	23		6.3929	1,1001	23	8.1429	1.4325	55
`	PISVIZ	0.4225	0.5040	23		1.8571	0.9315	23	1.9454	1,3131	55
	PI3V15	11.0357	5,1439	23		0.3214	0.4755	29	0.3750	0.4115	55
	XTIMINS	5.3757	2,3493	23		9.8214	5.0410	23	10.4255	5,0142	55
,	STIMINS	5.5321	1.6973	23		5.8494	2.2043	23	5.6227	2.2537	55
	STIMPASZ	0.9337	1.6301	23		5.8109	2.2924	29	5.7055	2.0051	55
£."	STIMIDLE	1.3419	0.9392	23		0.7502	0.5485 ,	29	0.8555	1.2437	55
	YDIFMATL	0.9233	0.2224	23		1.0225	0.9730	28	1.1823	0.9735	- 56
	XEDYRELB	20.8432	21.0151	23		0.9437	0,2220	23	0.9370	0,2223	55
	SLEADER	1.7016	0.5374	23	1	1.2049	29.2207	25	26.0225	25.75-5	55
	XPEDIMAT	24.8595	24.7791	23		.35:0	0.8:31	23	1.5055	0.7229	55
	SPEDYMAT	21.7633	13.5111	- 23		.07-3	29.6110	23	34.4654	23.6319	55
	SF3TIM3	17.5576	13.0715	23		3.7323	16.5459	23	25.2505	17,9313	55
	NPSTINS -	9.4113	7.1492	23		7.0502	11.6461	23	22.3099	13.1760	55
	SPSTIMS .	10.9014	7.3021.	23		3.6237	7.3540	. 23	9.0177	7.2021	56
	XTTIMI .	8.4651	1.0300	23		2.2341	8.3131	23	, 11.5573	7.7815	55.
	STTINE	1.6552	0.8434	23		. 8160	1.3745	23	8.1405	1.2151	55
	STTIMS	1.4713	1.4757	23	2	. 2512	0,8529	23	1.9647	0.8945	55
	XTTIM7	0.0265	0.1417			.6224	1.5+35	23-	1.5469	1.4939	55
	KFO3NEG	119.0325	93.4253	23 **		1.2411	0.4235	23	0.1339	0.3310	55
	XCCOPER	0.1935	0,3123	23	. 63	1900	44.3153	23	91.1114	77.1655	55
	CHX73	52.5755		23		.1071	0.1303	25	0.1503	0.2412	55
	CH173	48.4379	3 5 9 7 7	23		.4671	6.0731	23	59.0215	8.1302	55
	_C2073	49.9435	3.2991	27	- 5 9	.6979	4.3550	10 - 🛸	51.4512	6.1837	37
•	CT07773	47.6532	4,6537	23		. 9579	6.4135	23	56.4557	8.6153	55
		71.0375	4.1332	28	63	. 4714	6.2031	23	56.5573	8.7093	56
									-		

79.

٩

83

۰.

B.2 Legend for Classroom-level Analysis Variables

•	VAR001	ID	1	
	VAR002	RESPONDENT TYPE	CPISV07	PRINCIPAL TREATMENT IMPLEMENTATION SCALE
•	VAR003	OBSERVER NUMBER	PISVOB	PRIN PERCEIVED SALIENCE OF TREATMENT
			PISVI2	PRIN OPINION OF FUNDS ALLOCATION
•	VAR004	SCHOOL CODE	PISVI5	PRIN PROCESS DETERMINANTS SCALE
,	VAR005	RESPONDENT CODE	XTIMINS	IND OBS TIME IN INSTRUCTIONAL ROLES-ME
/	VAR006	GRADE	STIMINS	IND OBS TIME IN INSTRUCTIONAL ROLES-SD
			STIMPAS2	IND OBS TIME IN RECEPTIVE ROLES-SD
	GRADE		STIMIDLE	IND OBS TIME IDLE-SD
	VAR007	TREATMENT CODE	XDIFMATL	IND OBS NO OF MATERIALS USED-MEAN
	· .	1. SAT	XEDYRELB	IND OBS PERCENT OF EDY MATLS-MEAN
		2. CONC	SLEADER	IND DBS SCALED GROUP LEADERSHIP-SD
œ	PSPAN	PROPORTION SPANISH SURNAME PUPILS	XPEDYMAT	TCH OBS PERCENT OF EDY MATLS-MEAN
	PNEGR	PROPORTION BLACK PUPILS	SPEDYMAT	TCH DES PERCENT OF EDY MATLS-SD
	PCAUC	PROPORTION CAUCASIAN PUPILS	SPSTIM3	TCHR DES PCT PUPIL TIME ACTIVE-SD
7	GEN2	MEAN PUPIL GENDER OF TCHRS 78 CLASS	XPSTIM5	TCH OBS PCT PUPIL TIME RECEPTIVE-MEAN
	AGE2	MEAN PUPIL AGE IN MONTHS OF TCHRS 78 CLASS	SPSTIM5	TCH DBS PCT PUPIL TIME RECEPTIVE-SD
۰.	STAG	UNIT OF INSTRUCTION	XTTİMI	TCH OBS TCHR TIME INSTRUCTIONAL RCLES-MEAN
	· · ·	L. TOTAL CLASS 2. Staggered reading	STTIMI	TCH OBS TCHR TIME INSTRUCTIONAL ROLES-SD
	WK77	MEAN WORD KNOWLEDGE SCORE SPRING 77	STTIM6	TCH OBS TCHR TIME HELP PUPIL ACT-SD
	WA77	MEAN/WORD ANALYSIS SCORE SPRING 77	XTTIM7	TCH OBSTCHR TIME IDLE-MEAN
•	RD77	MEAN READING SCORE SPRING 77	XPOSNEG	SUPPORTIVE X NONSUPPORTIVE AFFECT-MEAN
	TOTR77	MEAN TOTAL READING SCORE SPRING 77	XCOOPER	COOPERATIVE BEHAVIOR-MEAN
	TISV01	TEACHER EXPERIENCE SCALE	CWK78	MEAN WORD KNOWLEDGE SCORE SPRING 78
	CTISV07	TEACHER TREATMENT IMPLEMENTATION SCALE	CWA78	MEAN WORD ANALYSIS SCORE SPRING 78
	TISVO8	TCHR PERCEIVED SALIENCE OF TREATMENT	CRD78	MEAN READING SCORE SPRING 78
•	TISV15	TCHR PROCESS DETERMINANTS SCALE	CTOTR78	MEAN TOTAL READING SCORE SPRING 78
*	PISVOI	PRINCIPAL EXPERIENCE SCALE		
_	0			A is a set of the s
Fulls		83	y an de la	86
. —	,			

B.3 Pupil-Level Analysis Variables

			Grade 2			Grade 4	
5	V-3149LI	N 7 1 N	STANJARU DEV	CASES	1 . 1	うてよいりょうJ D E V	C4573
1	インゴンシン	1, 25-1	-/ 0.5010	113			
	V47 J05	2, 1, 1, 1, 1, 1	0,0	113		J 5 7 2 5	132
	V12J05	1, 2 3 3 1	5012	113	1,00) >	ت ۵ د ت	102
	ソンマロウア しょう	1,5579	0, 3312		しょうウンフ	10302E	102
	V:=JOJ \	41,7573	11,1637	113	1日元(5月37)	······································	102
	V13010	30, 2145		117	-0,7:22	- 10,9375	
	V47011 N	=), 2 = = =	5,2352	107	ちゃ アリアウ	10,0444	101
	VA7)12	37, 57, 27, 2	1294334	105	K 3., 2 + 1 +	10,2702	37
	Y12,013	= 3, 2013	y0235+	117	77,1523	10,00000	
	V12014	A 7, 1 5= 1	5.15.79	107			1.01
	V43115	50,2752	7,2303	103),)	11,7351	.e - 9,
			10379	105	÷ • • • • • • • • •	0	ຸ່ປ
	XSTIC	1.0000	0.0	Э	1,2200	12,0728	15
	X ED Y	1.5-10	し,ユアミユ	113	1.5	250	7
	XNRDLIS	1,4575	0,5003	113		J,477	102
· · -	× X39.2512	1.5347	0.7771	113		5013	1.01
	V323512	-1063317	5.3225	- 113 -	1011275	05 2073	191
		1,3743	4357	113	1, 17, 5		1.12
	5-141 NS	25,7924	7.9023	113		235うじ4	175
	STIMULN	3,7001	3,0276	113		225054	102
	STI MACT	10,3370	3.3376	113	3,0175	J 2 7 0 6 -	102
	STI 47 SP	13,3335	0,5202	113	12, 2255	13573	102
	571 VP ± 52	1,9793	J. 6743	113	14,2215	1121205	1.02
,	STIMIC	3,0424	++7323	113	15771	1, デユ デミ	152
÷ ,	KUSEDY	0,2101.	0,2563	113	2,13-1	ようさいちょ	1,12,
	VU370Y	3, 2555	J.2 574	113	1,2450	しょじこそう	1,52
	XOIFMATE	3, 7211	J.30.37 /	113	7,2-17	J. 2011	
	VOIFMATL	0,4533	0,3137		0,7771	0,0-35	1 32
	KLEADER	3,5307	1,2697	113	2. 2. 5.33	J.3340	112
	VETADER	1,4007	112371	117	じょうちゃしつ いい	4.4162	
	D // K	11,1225	<u>0,7430</u>	117	1,7273	0,73:4	· 101
	243	7.3700	105302	103	5,1224	3,5325	B
	5,7373	11,2:02	7,2068	100	2,2		, 7 3
		1192772	7.9532	104	4,2447		3
	;					5,05.7	·75

•

18

÷.

7 ..

87

· · ·

5

83

È.

B.4 Legend for Pupil-Level Analysis Variables

i e

• • •	•		
	VAROO1	SCHOOL CODE	£
•	VAROO2	PUPIL ID	
,	VAROO3	TEACHER CODE	Ň
	VAR004	TREATMENT 1=CONS. 2=SAT.	0
•	VAROO5	GRADE	
/	VAR006	GENDER 1=MALE, 2=FEMALE	
	VAR007	ETHNICITY 1=SPN 2=NGR 3=CAU 4=OTH	L ·
	VAROO8	INDIV OBS FLAG	
	VAROO9	WORD KNOWLDGE 77	
	VAR010	WORD ANALYSIS 77	
	VAR011	READING COMP 78	
X .	VAR012	TOTAL READING 77	•
	VARO13	WORD KNOWLDGE 78	
	VARO14	WORD ANALYSIS 78	
	VAR015	TOTAL READING 78	•
	FALL	FALL 77 AS PRESCORE FLAG 1=YES	
	XSTAG	AVERAGE STAGGERED RDG., STA =2	
	VSTAG	SD STAGGERED VS TOTAL CLASS RDG	
	XEDY	EDY STATUS1=YES 2=NO	
 ,,	XNROLES	MEAN NO OF ROLES CODED	
	VNROLES	SD NO OF ROLES CODED	1
-	XGRPSIZ	AVERAGE INSTR GROUP SIZE	
	VGRPSIZ	VARIABILITY OF INSTRU GROUP SIZE	
•	STIMINS	TOTAL MINUTES INSTR TIME	
	STIMNIN	TOTAL MINUTES NONINSTR TIME	ς.
	STIMACT	TOTAL MINUTES PUPIL ACTIVE ROLES	
•	STIMRSR	TOTAL MINS PUPIL RESPONDING ROLES	
	STIMPAS2	TOTAL MINS PUPIL PASSIVE ROLES	
	STIMIDL	TOTAL MINUTES PUPIL IDLE	•
2	XUSEDY	AVERAGE USE OF EDY MATERIALS	
aj di	VUSEDY	VARIABILITY IN USE OF EDY MATLS	د. افغان
	XDIFMATL	AVERAGE USE OF DIFFERENT MATLS	
	VDIFMATL	VARIABILITY IN USE OF DIFF MATLS	• •
· ·	XLEADER	TYPICAL ROLE-GROUP-LEADER	
	VLEADER	VARIABILITY IN ROLE-GROUP LEADER	
	DWK	DIFFERENTIAL WORD KNOWLEDGE	
	DWA	DIFFERENTIAL WORD ANALYSIS	t.
· .	DTOTR	DIFFERENTIAL TOTAL READING	
FRIC	~ "		83.7
Full Text Provided by Ef		82.	- -

APPENDIX C: INTERCORRELATION MATRIX FOR CLASSROOM-LEVEL ANALYSIS VARIABLES 8

90

83.



Appendix C: Intercorrelation Matrix for Classroom-level Analysis Variables

	•						•								÷ .	•				
	•	V13037	. PS PAN	· Philip	PCAUC	GENZ	AGEZ	STAG	K X77	HA77	R077	TOTR77	TISVOI	CTISV07	TISVCO	TISVIS	PISVOT	CPISV07	PISVCO	
	V17:37	55.	55.	55.	56.	56.	56.	55.	56.	37.	56.	55.	56.							
·	P374:1	0.04315	55.	55.	56.	55.	56.	56.		2.4 37.	55.	56.	55.	56.		56.	56.	55.	55.	
	FUILOR	-0.06732		. 55.	56.	55.	55.	55.	56.	37.			55.	56.	. 56.	55.	56.	55.	56.	
	FCILC	0.05237		-0.31714	55.	55.	55.	55.	56.	37.		56.		55.	55.	55.	55.	55.	55.	
•	GE7:2	-0.93310	0.15552	0.00253	-0.15333	55.	55.	55.	55.	37.	55.	55.	55.	56.	56.	55.	55.	56.	55.	
	AGIZ	0.15727	0.03333	0.00319	-0.11930	0.01755	55.	55.	56.	37.	55.	56.	55.	.55.	56.	55.	56.	55.	55.	
	5123	0.01213	-0.02227	-0.03129	0.15275			55.	56.	37.	55.	56.	55.	55.	55.	55.	55.	56.	55.	
	_ 5 .77	0.19324	-0.11242	0.12475	0.09302	-0.25399	0.27157	0.03571	56.	- 37.	55.	55.	55.	55.	55.	56.	- 56 .	56.	55.	
	14A77	0.13772	-0.11935	0.03527		-0.45715	0.05722	0.13357	0.93557	37.	37.	- 37.	55.	55.	55.	. 55.	56.	56.	55.	
	8377	0.22357		0.11255		-0.27193	0.33573	0.00358	0.96145	0.91171	55.		37.	37.	37.	37.	37.	37.	37.	
	10:377		-0.11431	0.12330		-0.25150	0.32162	0.01335	0.93511	0.94250		56.	55.	55.	55.	55.	56.	56.	55.	
	TI3738	-0.14323	-0.34437	-0.04335	0.21120		0.13497	-0.02374	0.07252	0.05913	0.97174	55.	55.	55.	56.	55.	55.	55.	55.	
	CTISV97	0.07712			0.03747		0.14570	0.04554	0.00435	-0.21414	0.10551	0.03.37	55.	56.	55.	55.	55.	55.	55.	
	TISV23	0.07464	-0.05333	-0.17543	0.03531	0.07361	0.13585	-0.23237	-0.22493	-0.21075		-0.01325	-0.12531	55.	56.	55.	55.	55.	55.	
	TI5715	0.10743		-0.10555	-0.03013			0.10536	-0.30652	-0.07182		-0.18779		-0.053:5	55.	55.	54		55.	
	PI5731		-0.34407	-0.01535	0.21120	0.07122	0.13477	-0.02374		0.05913		-0.28554		-0.05350	0.08520	56.	55.	55.	∱ 55.	
	C7I3737	0.07912	-0.03523	-0.24025	0.07747		0.14570	0.04554	0:00486	-0.21414	0.10551	0.03/37		-0.12831	0.02145	0.06132	55.	55.	55.	
	PI5123	0.07464	-0.05333	-0.17543	0.03531	0.07351	0.13535					-0.01325		1.0000	-0.05355	-0.05350	-0.12331	55.	55.	
	PISVIZ	-0.08335	0.02943	0.03247	-0.01337	0.11335	0.04517	0.03754			-0.15135	-0.18770		-0.05335	1.00000	0.03520	0.02148	-0.05335	55.	
	P13715	0.10943	0.125:5	-0.10555		0.10554	-0.03347	0.10525	-9.30552	-0.07132		-0.15330		-0.10373		0.13177	0.12918	-0.10373	-0.19426	
	XTIMINS	0.21035	0.09472		-0.05350	-0.16201	-0.19517	-0.16723		0,13026		-0.23554		-0.45350	0.05520	1.0000		-0.05350	0.03520	
	STINCES	0.25723	0.05057		-0.03577	0.02337	-0.05505	-0.31835	0.18253	0.18133		0.14157		-0.01695		0.07309		-0.01693	-0.05425	
	51122452	0.17771	0.02443	0.15512	-0.10529	-0.04719	0.05843	-0.15141	-0.03502	0.01587	0.16240 0.00177	0.17331		-0,01910		0.05139		-0.01910	-0.07555	
	STIMICLE	-0.13502	0.05220	0.21233	-0.22249	0.07533	0.05755	-0.19245	-0.20402	-0.14700		-0.01567	-0.01650		0.10718		-0.01650	-0.143:5	S. 11 - S. 1	
	XOTEMATE	-0.01557	0.11343		-0.11473		-0.24103	-0.07333	0.07118	0,15342	-0.13771 0.11932	-0.19702	-0.02035		0.13153	-0.05195	-0.02058	-0.04940	0.13153 /	
	XEDIRELB -		-0.05214	3.21035	-0.00453	0.00855	0.16469	0.01730	0.21068	0.16725	0.16230	0.09182						-0.11190	-0.15935 1	
	SLEACER	-0.23534		0.65545	0.13772	-0.00843	-0.34572	0.05745	0.04333	-0.03345		0.18523	-0.04344			0.05039	-0.04344	0.13857	-0.12253	
	XFEDIMAT		-0.13475	0.29372	0.06951	0.05115	0.23715	0.03125	0.40755	0.19937	0.00200		-0.07505			-0.03929			-0,17437	
•	SFEDIMAT	-0.00553	0.02422	-0.03329	-0.04321	0.01137	0,17250	0.14004	0.25933	0.17831	0.42529 0.25747		0.03971			0.13539	0.03971		-0.10825	
	SF57:73	0.24015	0.17751		-0.14124		0.26935	0.21914	0.15558	0.15537	0.17737	0.25541		-0.05097	0.07110	-0.03261		-0.06297	0.07110	
	XASTINS		-0.14532	0.13394	-0.01671	-0.10126	0.27352	-0.17539	-0.05574	0.03735		0.17275	-0.05452		-0.04078	0.15344	-0.05452		-0.04075	
	5737145	0.07723	-0.11547	0.14754	-0.04432	-0.06559	0.22350	-0.21720	0.07812	0.077-5	0.02478	-0.01075	0.05793		0.05542	0.07350	0.05795	0.17634	0.09542	
	STTING	-0.03277	C.12575	-0.23182	-0.03429	0.13553	-0.45526	0.14191	-0.19671	-0.20164	0.14430	0.12161	0.07273		, 0.12224	0.10337	0.07273	0.09824	0.12224	
		-0.01215		0.24131	-0.03293		0.35901	-0.06494	0.26031	0.24114		-0.22920	-0,17579		0.08725	0.17007	-0.17579	-0.21697	0.03725	
	STTINS	0.31923	.0.11575				0.16342	0.25305	-0.00745	-0.03471	0.26379	0.27354	0.03723	0.23591	0.04521	-0.01955	0.05723	0.23591	0.04521	
	XTTIN		-0.10325	0.23725		-0.12614		-0.21043	0.32707		0.00755	0.00032							-0.03327	
	XPCS	0.10300	-0.26152	0.24148	· · · · ·		-0.24988			0.32175	0.32641	0.32636	0.15574		-0.13432		0.15574	0.00205	-0.13432	
	XCCC. 'S	0.29554	0.05777 _	-0.16574	0.05350		0.10335			-0.22424	-0.24974	-0.22876	-0.13597	0.17304			-0.13577	0.17334	0.10374	
	CLX73	0.16718	-0.07354	-0.02943		-0.13258		-0.19378	-0.06708	0.07293	-0.05013	-0.05128		-0.05450	0.14612		-0.27277	-0.05450	0.13512	
	73 شنت	0.04795	0.14159	-0.31935	0.04733		0.26855		0.70407	0.49613	0.74575	0.73697	0.13242		-0.07571		54221.0	56751.0	-0.07571	
	CF373	0.13055		0.02180	0.10115		0.38114		0.58538	0.45881	0.60948	0.61560	0.25345	0.11274			0.25345	0.11274	0.03732	
	CT01778	0.12331	-0.07370	0.04429			0.39220		0.69743	0.45039	0.75356	0.73554	0.19596		-0.12233		0.17375	0.03+21	-0.12233	•
				-		-0.00304	0.37020	-0.17705	0.70663	0.46036	0.74991	0.73504	0.19242	0.13196	~0.14573	-0.12554	0.19242	0.13116	-0.14573	
													,							

Appendix C (continued)

95

Full Taxt Provided by ERIC

5		•		XT IMDS	STIMINS	STIMPAS2	STINIOLE	XUIFMATE	XEDTRELS	SLEADER	XFEDIMAT	SPEDTHAT	525TIM3	XPSTIMS	SP5TIN5	XTTIN	STTIN	STTINS	XTT187
		PISV12	P15V15	AL 1.121.5	212						.			55.	55.	55.	55.	55.	55.
	VA3337	56.	56.	56.	. 54.	55.	56.	56.	56.	56.	55.	55.	55. 56.	55.	56.	55.	55.	55.	55.
•	#571N	54.	56.	55.	55.	56.	55.	55.	55.	55. 55.	55.	55.	56.	55.	55.	55.	55.	55.	55.
•	PHESR	55.	55. '	55.	56.	55.	56.	56.	56. 56.	55.	55.	55.	55.	56.	55.	56.	55.	- 55.	55.
	FCAUS .	55.	55.	55.	55.	55.	56.	55.	55.	56.	55.	55.	55.	55.	55.	55.	55.	55.	55.
	65112	563	56.	56.	56.	55.	56.	55.	56.	55.	55.	55.	56.	55.	55.	55.	_ 5à. ¿	56.	55.
•	ASEZ	35.	56.	55.	55.	56.	55.	55.	56.	55.	55.	55.	55.	55.	. 56 .	55.	55.	55.	55.
:	5713	56.	56.	55.	56.	55.	55.4	55.	56.	55.	× 55.	55.	55.	55.	56.	55.	55.	. 55.	55.
	SK 77	55.	56.	55.	55.	56.	37.	37.	37.	37.	37.	37.	37.	37.	37.	37. "	37.	37.	37.
	- HA77	37.	37.	374	37.	• 37.	55.	56.	55.	56.	55.	55.	56.	55.	55	55.	55.	55.	55.
1	E977 .	. 56.	56.	55.	55.	55.	55.	56.	56 .	56.	55.	55.	55.	56.	55.	55.	55.	55.	55. 55.
1	101377	55.	55.	56.	56.	55.	55.	55.	55.	56.	55.	55.	56.	55.	55.	56.	55.	55. 55.	55.
	TISVOI	55.	55.	55.	55. 55.	55.	55.	55.	56.	56.	56.	56.	56.	56.	55.	56.	55.	55.	55.
ŧ	CT13737	55.	56.	55.	55.	56.	56.	56.	55.	55.	56.	55.	55.	55.	55.	56.	55. 55.	55. 55.	50.
. :	- TISV:S	55.	55.	55.	56.	-56.	56.	55.	56.	. 56.	55.	56.	55.	55.	55.	56. 55.	55.	55.	55.
1	TISVIS	56.	55.	*	55.	56.	56.	55.	55.	55.	55.	55.	55.	55.	55.		55.	55.	55.
•	PIS'31	55.	56	55.	56.	56.	56.	55.	56 .	56.	55.	55.	55.	55.	55.	55. 56.	55.	55.	55.
	CPISVO7	55.	55.	56.	55.	56.	55.	56.	55.	55.	55.	56.	56.	56.	55. 55.	55.	55.	55.	55.
ä	PI5783 PI5712	55.	56. 55.	55.	55.	\$ 56.	55.	55.	55.	55.	55.	55.	55.	55.	55.	55.	55.	55.	55.
*; co	PISVIS	56. 0.13177	56.	55.	55.	56.	55.	55.	55.	55.	55.	55.	56.	55.	55.	55.	55.	55.	55.
, u	XTIMINS	-0.25-75	0.09337	55.	55.	56.	55.	55.	56.	56.	55.	55.	55.	55. 55.	55.	55.	55.	55.	52.
. i	ST17193	-0.24540	0.05130	0.82043	55.	55.	55.	55.	55.	55.	55.	56.	55. 56.	55.	55.	55.	55.	55.	17.
:	ST:#7152	-0.16589		0.31329	0.24573	- 55.	56.	56.	56.	. 55.	55.	55.	55.	55.	55.	55.	Sá.	55.	57.
• `	STIMICLE		-0.05195	-0.21231	0.12855		55.	55.	55.	56.	55. 55.	55.	55.	55.	55.	55.	55.	55.	,55.
•	XDIFFITL	-0.14192	0.07933	0.39349	0.25462	0.11547	-0.16873	55.	55. 56.	55.	55.	55.	55.	55.	55.	55.	55.	55.	55.
:	XEDIRELS .	0.05785	0.05039	0.19555		-0.12551			-0.18095	55. 55.	55.	55.	55.	55.	55.	55.	55.	55.	55.
	SLEACER	0.05165	-0.03929	0.14505		-0.05555	0.07656	0.05109	0.53919	-0.09327	55.	55.	55.	55.	55.	55.	55.	55.	55.
	"XFEGINAT	0.07653	ų.1353P	0.14827	0.12124	0.05322			0.12433	-0.18713	0.45523	55.	55.	55.	55.	55,.	56.	55.	55
	SPECYMAT	0.01620	-0.03251	-0.15051	-0.19530	0,05403	-0.24533			-0.32524	0.09775	0.07317	55.	55.	55.	55.	55.	55.	55.
	SFSTIM3	0.64026	0.15344	0.10156	0.00576	0.20697		. 0.07734		-0,15377	-0.00355	-0.03747	-0.05135	55.	55.	55.	55.	55. 55.	55. 55.
4	XFST173	-0.41504	0.07335	0.12533	0.13933	0.14635		0.10975		-0.11437	0.07237	0.02566	-0.07373	0.82572	55.	56.	55.	55.	55.
	SESTING	-0.32792	0.10337	0.19120		0.02177	-0.18187		-0.23805	0.03726	-0.04234	-0.01054	-0.29192	-0.35511	-0,19747	55.	56. 55.	55.	54.
	STTINI	-0.12224	0.17:07			-0.01054	0.05401		0.15025	-0.14575	0.10333	0.15675	0.37757	0.22531	0.19733	-0.75557 -0.01414	0.03181	55.	55.
	577:25	0.23553	-0.01555	0.02759			-0.14533	0.07459	0.11079	-0.32933	0.05533	0.04519	0.73921	-0.03357	-0.10061	-0.60139	0.47245	0.14532	55.
	311:37	0.17553	-0.25077				0.17377	0.18119		-0.26002	0.00032	-0.02753	0.23739	-0.00452	0.04615	0.05740	0.03327	0.00153	-0.53533
• ~	X711.13	-0.62335	0.01733	0.05123		9.53358		0.00104	0.15527	0.16505	0.065-5	0.14705	-0.07143	0.15459	-0.11671	-0.00071	0.07745	0.35455	-0.07273
	X121713	0.19448	0.16334	0.02319		0.27777		-0.01537	-0.13313	0.11625	-0.03123	0.03117		0.03237	0.12411	-0.39403	0.31323	0.13327	0.3*273
	C1473	-0.18511	-0.23225	0.27118	0.27400				0.24352	-0.14523	0.27239	0.03350	0.35431 0.42343		-0.04053	-0.11053	0.11357	0.23257	0.23312
· · · ·	- C2123	-0.25447		-0.04540					0.13775	-0.22105	.0.03150	0.03574	0.42343	0.05228	0.15218	-0.35233	0.32041	0.11535	0.3-512
	C-213	-0.16255	-0.17537	0.16757	0.20707	-0.03335	-0.20555	0.26422		-0.07770	Q.23783 0.30764	0.05078 0.05492	0.35501	0.06923		-0.40269	0.34227	0.12235	0.37153
	[C131375	-0.1603Z	-0.12554	0.19463	•0.23472	-0.03720	-0.20433	0.24101	0.21542	-0.11834	0.30704	0.03476	*******				13		

Фę

 9_{\pm}

(COUGTAGEA)	
-------------	--

	δ.	•		. •		
	XPOSHEG	XCCOFER	C1:X78	CHA78	CRD78	CTOTR78
	<u> </u>	55.	56.	37.	56.	56.
VL9007	56.		~55.	37.	55.	56.
PSPAN	55.	55.		37.	557	55.
PLIEGR	\$6.	55.	56.	37.	55.	56.
FEAUE	56.	55.	55.	37.	56.	55.
GE::2	55.	55.		37.	55.	55.
A312	55.	. 55.	55.	37.	55.	55.
STAG	. 55.	55.	55.	37.	55.	55.
WX77	55.	53.	55.	30.	37.	37.
KA77	37.	37.	37.	37.	55.	55.
2077	5\$.	. 55.	55.		56.	55.
TOTA77	-56.	55.	55.	37.	55.	55.
TIS:01	55.	55.	55.	37.	55.	55.
CT15V07	55.	. 55 .	55.	37.	56.	56.
TISVCS	56	55.	56.	. 37.	55.	55.
TISVIS	55.	55.	55.	37.		56.
	56	55.	55.	. 37.	56.	56.
FISTON	55	55.	55.	37.	55.	55.
CPISV07	55.	56.	55.		55.	55.
PISVOS	55.	55.	55.	- 37.	50.	
P:5%12	55.	55.	55.	. 37.	56.	55.
PISV15	55.	55.	55.	37.	55.	55.
XTININS		55.	, · 55.		, 55.	56.
Stimins	56.	55.	· -		, i 56.	55.
STIMPLSZ	55.	55.			50.	56.
STIMICLE	55.	55.			56.	55.
XCIFHATL	55.	55.			56.	56.
REDIRELB	55.	55	·	•	55.	55.
SCEASER	55.	55			56.	56
XPEDIMAT	55.			•	56.	56.
SPEDIMAT	56.	56		-	55.	
SPSTIN3	56.	56 55		•	55.	55
XPSTINS	56.	55	-	•	56.	56
SPSTINS	55.		-	•	55.	-55
XTTIH	55.	55	•	•	56.	56
STTIM	55.	. 56	• •	•	56.	- 55
STTIN5	. 55.	55		•		
XTTIN7	55.	55		• • • • •		
XFCENES	55.	/ 55		•	- 1 T 17	
XCCOPER.	0.05523	- 55				
C:	-0.43531	-0.14:5				·
C1173	-0.31173	0-0113	3 0.8372		•	·
CE378	-0.43214					
CTOTR78	-0.40412	-0.1574	2 0.9494	0.81283	0,9392(, 50

95

86.

7

-

APPENDIX D: SUMMARY OF REGRESSION ANALYSIS

93

87.

D.1 ON SECOND GRADE PUPIL

1

D.2 ON FOURTH GRADE PUPIL

a

APPENDIX D D.1 SUMMARY OF REGRESSION ANALYSIS

SPENDENT V	VARIABLE VARUIS ORC KNOWLDGE 7	SUMM	AFY TABLE		- 	e 16	₽=+ 4
	· ·	MULTIPLE P			5' HOLE 9	6.853755	.0 - 31 = 0 1
	EDY STATUS ., I=YES 2=NU BORD KND4LCGF 77	0.58837 0.63166	0.34577 C-36000	0,34477 0,05223	C, 59437	0.2932324 30.50110	0 , T320 I
-DEPENDENT	VARIABLE VAROIA MORD ANALYSIS	78					
S .		· .	•	•,	·	ал на Стала стала ста Стала стала	
ġ./ ·	EDY STATUS 1=YES 2=NJ #DRD ANLYSIS 77	0,62747 0,63893	0.37372	0,39372 0.090-0	0,42747 0,53153	6.257242 0.335±344 25.30524	0,12320 0,124331
		-				•	
CEPENDENT	VARIABLE VAROLI READING COMP 70	3				•	
•	· · · · ·	×				7.893664	0,37953
· · · ·	EDY STATUS •• 1 =YES 2=NU TYPICAL ROLE-GROUP LEADER FALL 77 AS PRESCORE FLAG 1-YES TOTAL MINUTES PUPIL IDLE TOTAL FEADING 77	0.63572 U.66455 0.67960 0.69224 0.73114	0.40414 0.44142 0.44186 0.47319 0.53457	0,40414 0,03749 0,02023 0,01733 C,05529	0,63542 -0,17046 C+12463 0,14838 C+66389 C+66389	-1.050081 5.522303 0.2255274 0.4071502 25.5079	-0,12779 0,14109 0,10242 0,34039
DEPENDENT	VARIABLE VARDIS TOTAL READING	78			-		
-			· · ·		н а		
۰. ۲. ۲. ۲. ۲. ۲. ۲. ۲. ۲. ۲. ۲. ۲. ۲. ۲.	EDY STATUS 1=YES 2=NU TYPICAL ROLE-GROUP LEADER FALL 77 AS PRESCORE FLAG 1-YES TOTAL MINUTES PUPIL IDLE TOTAL READING 77	0;63598 0.65807 0.63558 0.63558 0.63526 0.73257	0.40149 0.44632 0.47016 0.43756 0.53710	0,40449 0,04184 0.02384 0,01741 C,04953	0,63500 -0,19139 C-14070 C-14777 C,65076	47.641990 -1.051707 5.628354 0.2135537 0.3575424 27.93341	0,33444 -0,13793 0,13484 0,13445 0,71054
3		Í.	1				
		1	•				
ан Ф иллан С		• • •					
	· · · ·	1					

APPENDIX D D.2 SUMMARY OF REGRESSION ANALYSIS

REGRESSIAN **T** 1 PLE

÷

• :

DEPENDENT VARIAELS..

11.4

VYS 213 #ORD KND#LDGE 78

SUMMARY TARLE

			-	MULTIPLE R	R SOULPE	RSC CHANGE	6.Abîe p	, ę	9774
•	EDY STATUS - Total Minuts Total ² Mins F #080 Knowlog	S NONINST		0+67214 0+63619 0+70203 0+72712	0,45178 0,47034 0,49295 0,52371	0,43179 C,01309 0,02179 0,03525	C • 67714 C • 30463 • C • 08937 9 • 57793	10,39639 -0,5822759 -0,1107975 0,3772592 30,24544	1,32541 -9,1435 -9,7335 -9,74739
DEPENDENT	VAGIABLE	VARUII	READING COMP 78	•			•	х 	
	`~	•		. /					•

EDY STATUS ++ I=YES 2=+0 Average Staggired Nu0 ++ Stag#2 Total Reading 77 Total Minutes Nuninsta Time	0,58263 0,60195 0,65030 0,65635	0+33945 0-36234 0+44669	C-33545 D-02255 C:05055 D-C2390	0-58243 -0-16271 0-62614 -C-01516	6.552222 -3.575215 0.5233357 -0.5647631 25.11120	0 - 24 4 4 0 - 0 - 12 7 4 7 0 - 14 5 9 7 - 0 - 15 9 7
--	--	-------------------------------	--	--	--	--

0.41824 0.41924 C.47022 0.02201 0.57901 C.40222

DEPENDENT VARIABLESS

TOTAL READING TO

۰.

\$, EDY STATUS \$\$ 1=YES 2=NU TOTAL MINUTES NUNINSTR TIME TOTAL READING 77

95

VAAU S

0,55751 0.63578 0.76043

· . .

*

U

0,66761

------et 6-73213

5,533544 -0,5745474 0,6231995

13.60414

\$_C

0,12215

.

 $10_{.1}$

APPENDIX E: REGRESSION RESULTS WITHOUT EDY STATUS

E.1 SECOND GRADE

E.2 FOURTH GRADE

101

<u>,</u>

	. · · ·	· · · ·			
APPENDIX E: Second-	Grade Regessio	n Results wit	hout EDY Statu	S .	
E.1			•		
· · · · · · · · · · · · · · · · · · ·	•	, ,	_ 8 ,		· .
	LTIPLE	9 E G 9	T S S L O N		
ELEN VALUE ADED KNOILOGE 7	3				
		MACY TABLE			
	MULTIPLE R		RED CHANGE	STVDLE S	a
L MINS PUPIL RESPONDING FOLTS KND/LOGT 77	0,17407	0.03030	0 · 0 30 37 C · 3225 3	-0,17407	-0.1220642 0.5051107
				,	- 33.63573
	1				•
SLI., VAROLA AJRD ANALYSIS 78		•		• •	
	୍ୟ		•	. •	
					-4.051579
ABILITY IN USE OF DIFF MATLS AGE USE OF EDY MATEHIALS	0 • 1 7-37 4 0 • 255 3 1	0.03195 0.05519	C,03175 0.03323	- C, 17974 C, 15059	3.841425
L WINS PUFIL PASSIVE ROLES	0.30513 0.61372	0,07371¥ 0,33281	0,02953 C,29910	0-10412 Cr57150	0,1739,479 0,5365009
ANLYSIS 77 L MINS PUPIL RESPONDING ROLES	0.63419	0.40220	0,01,030	-0.120-0	-0.1333463 30.19373
	1		×.		
	_			1	1 ^
SLT . VARULI READING COMP 78	e e e e e e e e e e e e e e e e e e e			۰ ۹	• /
	·		, f	· • • • • •	
	<i>a</i> .		· .	، من ^{مر} با مر من من من من من من من من من من من من من	the many
AGE UST OF LOY HATERIALS	0,17377 0,65429	C,03710 0,43129	0:03017	0.17277	1 • 1 30 247
LE TRACING 77 Cal Pili-Jruup Leader	0,67572	0,45921	0.01493	1	-1.073344 25.19371
	N				2341-01-
ALF VAROIS TOTAL READING 7	õ		·		
	Q				
	•				
AGE USE OF EDY MATERIALS	0.20350	0×04158 0×0≠925	0.02772	C,20270 -0,19179	2.224332
CEL ROLE-GREUP LEACER	0,26324 0.67471	0,0,0,0,20	0.32620	C-65676	0.6670910 27.50572
ERIC					
 ANITARX POTIDIDE OF AGE. 	•				

	- · · ·	· ·		· - · ·	•	, •	•
-	APPENDIX E:	Fourth-Grade R	legression Res	ults without E	DY Status		
- X	E.2		•		•		
			•	· · ·			
	•			**************************************	•		Same March
	· · ·		•	• •	÷ •		
	· · ·		0 = 6 = 3	5 5 1) N		* * * * * * *	
	*******	LTIPLE				۰.	-
ADAEL T., VARUIJ	AGRO KNOWLOGE 7		10. The second s	· ·			
		SUM	1257 TAPL=			- · ·	
· · · · · ·		MULTIPLE R	2 501140=	PSG CHANGE	CINDLE Q .	ç	
		0,17097	0,03649	0.03513	0,190=9	4.339102 0.7485045	-
VERAGE USE OF DIFFEREN	T MATLS /	0.72121	Cn 52014	0,47365	0.7004	15.92105	
DED KNOWLOGT 77				^		-	
·				•	۰ ک ر ا		, 1
	•	* · · · · · · · · · · · · · · · · · · ·			· *		
	READING COMP 7	,	•	* ****	•		
OILELE VARUII	READING COMP TO	-	12				
· •					and the second second	- *	
. * /	÷		· · ·	0-03735	0-10-21	2.709632	
	NT MATLS	0.19325	0-03735	0,03530	C-17-14 C-60-56	5.276313	0
WERIGE USE OF DIFFERE VARIABILITY IN USE OF,	UIFE MATLS	0.25763	0, = 07 5 4	0.43513	- C+ OCF 23	-0.0515349 14.24549	
TOTAL READING 77	7146	0,72557	0-=2270	0.0		1402-0.	2
TOTAL PIROF AUTOM	•	•		:		· · ·	2
	1			, 		د	
· · · · · · · · · · · · · · · · · · ·	TOTAL READING 7	4	- · ·	• /			
READERS VARUES	IUTAL ALADING P	- (ر					•
		7	•				
	1				C-21716	4.005 jao	
AVERING TUST OF DIFFEREN	NT MATLS	0,21716	0, 0, 1715 0, 0, 2332	0.01714	9,17512	5.716744	
LARTABILITY IN USC DE L	DIFE MATLS	0,27049 U,73978	0,62217	0, 53779	C. 77710	0:3004735	•
「月下五日」 3月 3月 3日 - 77 「月下五日」 M E N日二月 5 - 13月 (M 2 T 4)		0, AJJ70 0, 1128	0,63918	0.02117	-0-000-0	3,071774 5, 224040	ана. С
AHING NO OF 77635-CDJLJ		0,32204	0167575	0.01757	0, 02 - 75	5,372810	
VERAGE USE OF LOY HATE							
			\sum				. 1

۴.

10.1

Full Taxt Provided by EBIC

APPENDIX F: FOUR-WAY ANALYSIS OF COVARIANCE FOR SECOND GRADE PUPILS

F.1 WORD KNOWLEDGE - 1978 F.2 WORD ANALYSIS - 1978

F.3 READING COMPREHENSION - 1978

F.4 TOTAL READING

- **1978** 1

	\$		
A-XAY ANCOVAR PEL DATH NE GRADE 2 AN IVAS . , DISTEC FELT POVA (CUTATION -		in the second second second second second second second second second second second second second second second	¥ {
	CATE = U1/25/79) L Y 5.1-3 C F V A C		•
	ND KHUALDUE 76 LATMENT RECONCLOSESAT	-1 A N C H + + , + , + + + + + + + + + + + + +	
	1 STATUS 1=YES 2=NC 1923 2=307. 2=6191	"Ti -	
	HALCITY LESPA 24-464-346 TAL MINS PUPIL RESPONDED NO ANGALOUS 77	AU 14=DTH & NJ 9DL95	
		* * * * * * * * * * * * * * * * * * *	
SCURC - OF VARIATION	5 00 45 ES DF	SOUAFE E SEE	(e)
YATN SFFECTS VAF034	3594:515	613,817 11.833 0,007	
XEDY VAR005 VAR007	3232,263 1 15,053 1	3232,263 62,118 0,000 3232,263 62,118 0,000 15,053 0,282 0,552	
C)V 17 LATES	114,079 3	38,693 C,744 C, 525	•
9 STI 425P VA2007	6235267 2 005623 1 5265833 1	314,534 5,027 0,004 60,528 1,165 0,254 526,839 10-125 0.002	<u> </u>
2-437 INTERACTIONE Viedua Xecy	0663625 1.1 73.743	60,602 1,165 0,224 73,949 1,410 0,237	•
VA7004 V47004 VAH004 V47007 XFDY V4704	4.337 1	4, 397 0, 004 0, 777 51, 372 0, 007 0, 777	
XIDY VAC077 VAC05 VAP077	1210323 1 1220521 3 3120136 3	121,323 2-332 0,131 40,344 0,7FT 0,F05 104,062 2-000 0,120	
J-4AY INTERACTIONS VARIOUS	1013.647	202,773 3,957 0,003	•
V4F004 X5DY V4C007 V4P00% V4P006 V3F007 X5DY V1F006 V4F007	750 413 2 256: 313 1	205,412 3,552 C.C. 47,907 2,910 0,103 255,313 4,564 C.C.23	
TXPLAINED	425,331 1	425,931 P.1P4 C.CO5	
47. SIDUAL	4318-532 83		n

10 CASES (2, 5 PC + ALAE MISSING.

105

2

107

FUITERE Provided by ERIC

		F.1: FOUR-WAY	Y ANALYSIS OF	COVARIANCI	E FOR SECOND GI	ADE PUPILS
	•		анан сайтан ал ал ан ал ан ал ан ал ан ан ан ан ан ан ан ан ан ан ан ан ан	• •	· · ·	•
4-WAY ANCOVA	5 246 747	المعادية أتوجعه الم	AULIS MODEL	Ż		
5740-2 4ND F1L- PDV4	(Je-1. A72 2 2 2 2	TTTSCORES TTTN DATE = U	。 【/25/79】			
••		, 4 %	<i>ه</i> ر،			
· · · · · · · · ·	• # • ± 3	- IALY – SA - Y_A+A G⊡ta	• • • •	7 2 1 7		
· · · · ·	EA . A70001	THEAT AEAT	1=C JNC .2=54T	NC		• • •
	メニフィー・ マミニフロテ					
р • (9	V1=077		1=5PN 2=157	3=CAU 4=		
्रम् मामा के के बा			10 //	* = * *	* * * * * * *	* * * * * * *
1	-			5	٣	
- -			564 55		11 E M	TTONE -
ROUTER OF VA	=: 47		5007523	5F	201432.	
MATH EFFECTS		•	2437,105	ċ	414,515	
VAF024 XEDY	ı.		413124 20053427	1	49,154 2066,427	1,251 0,257 54-0053 C-C03
VAFOJS	*		0:046	i	0,046	0.001 0.372
V3'20')7	• •	•••	152,531	3	54,294	1.421 0,247
COVARIATES	. v) - .	552,323	1 -	552,323	14,450 0.000
VAP010			552,323	L	552,323	14, 1= 0 - 0.00
2- ALY INTERS	CTIONS		555 ef 05	111	12,035	0.454 0.500
	101		C3 570 243 233	1	· U , 340 24, 350	$9 \cdot 9 = 3 + 9 \cdot 1 = 1$ $9 \cdot 2 = 2 = 0 \cdot 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2$
	V12004		469577	2	. 23,332	5.411 5- =45
	VERODA		c 5 . 7.33	l	· 5,733	1.721 0.124
XEDY	V 4 7 0 9 7		9,215	3	3,099	0,041 C+070 0,345 0,757
V47006	VYSC07	•	45.231	3,	15-077	
AFETNI YAN-E	CTICUS	· · · ·	171.53		34, 312	0,958 0,127
V12304	XEDA	DUC SALVA	1051647	1	105,647	2.766 C.107
		V 1 - 007	2,605	2 .	0,205)1 DCT C-=42
		V 4 = UU 7	4,7,72	i	4	2-125 0-724
EXPLAINED		•	3415,217	د ع	164, 666	1,958 0.000
RESTOUAL		x	2703,174	75	33,200	
TOTAL		\$	03220 EV4 m		63,944	
•	2			•		•

\$

ð.

118 CASES WERE ERICESSED. 19 CASES (15,3 PCT) ALKE MISSING.

·105

. .



	F.1: FOUR-WAY	ANALYSIS OF	COVARIANC	E FOR SECOND GE	ADE PUPILS	
ANY ANCOVAS PEL DAR	A HAA CAVARI	MA ES MODEL	•		ξ	. '
GRADE 2 ANDVAS P	TSTECCAES			•		
* * * * * * * * * * *	NNALYSIS	and the second s	ARIA	N C E * * *	• * * • • *	-
VAC011 97 VA2004	READING (CO) TREATMENT	MP 78 1=CJNC,2=SA	.т			1
XEDY 44	CUY STATUS GENJER 2=01	•• 1=YES 2 UY• 2=GI9L	2=N0		· .	
VARDO7 WITH XLEADE	P. TYPICAL RUI	1=SPN 2=NGA Lifgroup Le	ADER	É0TH .	- ,	
CIMITS VAROLZ		TES PUPIL I ING 77	DLE			
* * * * * * * * * * * *	* * * * * * *	* * * * * *	* # # #	* * * * * *	* * * *, * *	•
	·	SUM OF	, in the second s	HEAN	F CE	E
SOURCE OF VARIATION		SQUARES	JF.	SCUARE	F CS/	c .
MAIN EFFECTS VAROO4		4867.043 50.297	6 1	811,174 50,277	15,456 0,00	ni j
XEDY V47005		3757.857 63.001	1	3767-859 68-001	71,701 0,00	· · · ·
VA = 007		336,047	3	112,015	2,134 0,10	
COVARIATES XLFADER		1078.305	3	359,435	6, 84 9 C, 00 2,456 0,12	1
STIMIDL VAR012	•	57.745 669.852	, İ L	57,746 569,P52	1,100 0,20 12,763 0.00	7
2-WAY INTERACTIONS	· · ·	345.527	11 "	85,957	1-639 0-10	•
VAR004 XFDY VAR004 VAR006		1,291		- 1,291 - 51,595	0,025 0,87	'÷
VAR004 VAR007 XEDY VAF005		143.057	2	71.528	1,343 0.26 C,267 0,61	2
XEDY VAR007 VAR005 VAR007	•	104.693	3	34,999	0,645 0,57 2,057 0,11	'5
J-WAY INTERACTIONS	· · · · · ·	199.234	* 5	39. 947	0.759 0.58	-, ∞ •2 -1
VACUUA XCDY	V 1 7 3 3 5 V 1 7 3 7 3 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7	14,740 110,±47	1	14,740	0,291 0,53	94
V17004 V47005	▼ 4 □ 00 7 ▼ 4 □ 00 7 ▼ 4 □ 00 7	15273	1	1,273	0,024 0,33 0,146 0,70	-
	V 4 1 007	7090.109 °	25	233,604	5-404 0-00	
	1		77	52, 491		9
PESIDUAL V		+041+230	4			
~0Ţ4L		11131.340	102	109.131		•

119 CASES WERE PROGRESSED. 15 CASES (12.7 PCT) HERE MISSING.

110

ERIC

. ę

	.1: : FOUR-WAY ANA	LYSIS OF COVARI	IANCE FOR SECOND G	RADE PUPILS
4-117 1100415 DEL 1111	CUVARIAULE	S MODEL 2	•	•
FILT POVA (CPTATI	IN DATE = 01/25	177)	•	1
V2~015 ∃Y V±₽004 XTCY V42015	4 L Y 5 [5 TOTAL AZADING INCATMENT L=CD 20Y STATUS 32 (NETA 2 = 60Y.	1=Y=5 2=10 2=10		* * * * * * * *
V42017 V17H XUSEDY X11107 V49012	170010177 1=57 4002000 J50 0F 7971000 0000005 70700 0000000 70700 0000000000	SEDY MATERIAL SCUP LEADER	U 4=0TH _S < * * * * * * * *	* * * * * * * *
			_	· · ·
SOUPCE OF MARIATION		UM DE UARES DE	1511N 50149 デ	
9419 REFECTS V 27004 KEDY V 22004 V 22004 V 22007	ا بر 3 ا ب	2,535 5 2,273 1 2,211 1 5,034 1 5,034 1 5,255 3	635,473 13,278 3318,611 49,984 51,955	15,514 C,000 0,301 0,395 75,107 0,000 1,035 C,300 1,955 C,144
COV 17 1 ATES XUSEDY KLEIDE? V1-012	1 15	6 273 3 5 400 1 5 1075 1 3 240 1	295.624 15,400 155,005 559,240	6,634 0.000 0,340 0,757 3.510 0.065 12,634 0.001
2-414 : NT1 24071 FHR /12004 XEDY V12004 V12007 V12004 V12007 XEDY V12007 V12005 V12007 V12005 V12007	ء 1 آ 6	5,75 11 0,556 1 0,203 1 3,0666 2 12,742 1 5,646 3 3,0574 3 3,0574 3	70+ 887 01 556 50: 403 56: 533 1: 942 22: 222 129: 525	1.666 9.112 0.013 0.11 1.367 0.246 1.270 0.294 0.024 0.434 0.024 0.434 0.024 0.434 0.024 0.434
VA=004 X=3Y VA VA=004 V(40006 VA	≂∪し≂ 3 ≓ ∪∪7 15 ≑ ∪∪7	5,567 5 Jo 10 1 7,733 2 2,272 1 1,023 1	233103 33307 723977 23272 413023	1. 0.37 0. 7.5 1. 70 0. 17 0. 0.5 0. 17 0. 5 0. 17 0. 23 0. 23
TXPL STATE	c1 7	4,934 25	247, 723	
-73:3046	ن ب ر ب ر	2.234 77	44,175	
- ?T 3L	95.9	7,090 102	-4,C=0	2

113 CASES WERE PROCESSED. 15 CASES (12,7 PCT) ALRE AlssinG.

Full Text Prov ed by ERIC 11i

P

3

-*/

.

.

Ξ. .

.....

APPENDIX G: DIFFERENCE SCORE ANALYSIS FOR SECOND GRADE PUPILS

\$

- G.1 WORD KNOWLEDGE
- G.2 WORD ANALYSIS

G.3 TOTAL READING

	G.1 DIFFERNCE SCORE ANALYSIS FOR SECOND GRADE PUPILS
4-414 ANCOVAS ON DJP!L 07105 2 PUDIL D114 FILE POV4 (C234	•
	NALYSIS OF VARIANCE*******
37 (133)) " XFOY (473)) (473)) (474) (474) (474) (174)	TH_ATHENT 1=CUNC+2=SAT EDY STATUS + 1=YTS 2=NC GLADER A =GUY: I =GIPL STHNICITY 1=SPN 2=NGR 3=CAU 4=DTH TUTAL MINUTLS NCKINSTR TIME

X			VE 7 /		51 CH 15
SJURT OF VARIATION	30% DF 500A=E5	0ª	5011A7 E	E	7 = F
MAIN EFFECTS VAROUA XEDY VAROUA VAROUA VAROUA	751,033 6,031 552;563 33,377 134,651	6 1 1 3	125,131 6,031 5,2,563 33,377 6,5,897	1,05A 0,05A 0,56C 0,56C 1.013	0,022 0,740 0,604 0,641 0,371
CTV 4 F L A TES 3T I 4N IN VA 701 2	14770 275 3230 205 10020 916	2 1 1	733,929 323,205 1002,316	11-533 5:040 15:650	0,000
2-4 AY INTERACTIONS VAROUA VARCOS VAROUA VARCOS VAROUA VAROU7 XEDY VAROU7 KEDY VAROU7 VAROUS VAROU7	5 470 515 1 40 00 75 1 35 520 1 5 90 E 0 5 3 55 4 25 1 5 50 2 4 5 2 3 1 , 77 3	1 1 1 2 1 3 3	77,055 1-0,005 13,520 54,03 38,465 55,092 77,252	1,207 2,154 0,211 1,321 1,341 0,440 1,204	0, 273 0, 143 0 - 447 0 - 271 0 - 271 0 - 273 0 - 447 0 - 273
3- #AY INTERACTIONS VARUUA XEDY VARUUR VARUUA XEDY VARUU7 VARUUA VARUUA VARUU7	1323,010 231.857 212.013 222.253 530,676	5 1 2 1	264,602 201,957 105,007 222,248 530,676	4-129 3,159 1,654 3.469 7,052	0,002 0,007 0,054 0,054 0,054
	+399+633	24	193,320	2-9-1	0,000

FXPLAINED	+399+633	24
F 75 1704L	2318.357	83
-)- _	.712,074	107

119 CASES WERE PORCESSED; 10 CASES (9,5 PIT) WERE MISSING.

99.

لي قر

113

54,077 71, 523

G.2 DIFFERENCE SCORE ANALYSIS FOR SECOND GRADE PUPILS

LE DOVA (1774)	T187N U4TE =			~ - + ~ +	· · · ·	* • •
エ # ☆☆ # # # # # # # # # # # D #☆ TY V1=003 X=TY V1=00 V1=00 V1=00 V1=00	ロビア STAT ユージンジョンジー ユージー おします デート おします	T 1=00HC+2=3X J3 35 1=YE3 2 ==UY+ 2=0IPL Y 1=5PN 2=M3R	= 10	· · ·		
CIC:C:V:TH:V:TH:V:T)	مرتمی م.۲۲ م ه ه ه ه	1315 77 + + = = = + = + =	• • • • •		• • • •	* *
JICE DE VARIATION	·	50% 07 5004FER	CF	5 CU AD E	F	त्राहरू ग् इन
111 JFFECTS - VA1004 XTOY VAT005 VAT005	. . .	13333227 50110 33553 113040 1540130	÷ 	JO. 555 6,110 3,559 11,045 56,043	C. PON D. 140 J. 054 C. 7PC 1. 447	0 - 23
VARIATES VARIATES		1475,313	 1 1	1675,313	17, 254 47, 254	0,)) 0-)(
- XAY TITIAL CTTTIS VARUUA XFDY VARUUA VARUUA VARUUA VARUUA XFDY VARUUA XFDY VARUUA VARUUA VARUUA XFDY VARUUA		202.321 0.237 24.337 44.473 53.733 53.733 45.231	11 1 2 1 3 3	18,93* 0,840 24,359 23,339 5,733 3,070 15,077	0-476 0-639 0-639 0-611 1-721 0-205	
• AY IN TTACTIONS VARUUA XEDY V4R004 XEDY V4R004 XEDY V4R004 V4R005 XEDY V4R006	V 3 7 00 5 V 4 7 00 7 V 4 7 00 7 V 4 7 00 7	171,551 2,555 2,555 0,275 4,773	5 2 4 1 2 7 1	J4, 312 105,614 1, 333 0-206 4, 293	0, 253 2, 764 0, 0, 5 0, 0, 5 0, 0, 5 5, 1, 2 5 , 1, 2 5	
REALNED		6237,524	23	97, 327	2.560	0.00
IS TOUAL		2903,231	75	34+200		
JT AL		5141,753	33	= 1, 037		

114

118 CASES WEEK BURCHSSLD; 18 CASES (15,3 PCT) MERE MISSING.

•



		APPENDIX G
	G.3 DIFFERENCE SCORE	ANALYSIS FOR SECOND GRADE PUPILS
	L DIFFERINCE SCORES	· · · · ·
12 17 2 PUPIL 24TA		•
	UNERS SFV	ARTANCEFFFFFFFFFF
47 V.1.1) 17 V.1.1)	The AT 42 AT 12 42040 42=34T	
> -) Y V (=) ()		
Vはつ00 WITH V4つ01	ETHNICLIY 1=SPN 2=444	3=CAU 4=7TH
· · ·	3.011.3=	
SCURCE OF VARIATION	3.104=17	DF
MATH FFFECTS	237,217	30,538 0.805 0.503 1 30,500 0,565 0,507
VARDA XEDY	303 297 31, 172	81,972 1,553 C,177 23,026 0,453 0,423
VA7005 VACUU7	140,20 140,20	3 46, 934 1.047 0.179
: 1VA=14TES	+ 13175737 1	1 1317.747 27.330 0,000
VA7012	1 5 1 7 5 7 4 7	
2- ANY INTERACTIONS		
175007 XEUA	39,974	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
VAROJA VAROJA KEDV VAROJA	0,015	
V1=006 V1=007	- 5 2 ,	
3-44Y INTERACTIONS	4523725	5 21,545 2.005 0,074
VAR004 XEDY VAR004 XEDY	VARUUC 24.+01 VARUUF 175.202	2 37,601 1.004 0,141
VARODA VAROS	V4C/007. 49373 V3⊐007 51555	1 4,373 C,CCT 0,779 1 51,565 1-300 0-241
	2027.404	23 122,731 2.74 0.000
L X PL NINE D		
4 3 5 1 1 UAL		103 51,751
-3- AL	ງ 3 ຕໍ່ບໍ ໍ 4 ປ2	
118 CARES #505	····	
14 64525 (11,	()(T) ★171 (415) *(0)	•
	<u>.</u>	
· · · · ·	$\sim 10^{-10}$	1

Ē

.

FUILTERE ProvIded by ERIC

5

115

٠,

•

•

ť

APPENDIX H: FOUR-WAY ANALYSIS OF COVARIANCE FOR FOURTH GRADE PUPILS

 11_{0}

71

H.1 WORD KNOWLEDGE - 1978

H.2 READING COMPREHENSION - 1978

H.3 TOTAL READING - 1978

	LÜVA I AE TE = UI/2 KNJALUU TATUS TATUS TATUS TATUS TATUS TATUS TATUS TATUS TATUS TATUS TATUS TATUS	LIS MCDE 25/79) 0.F 78 CJNC+2=S 1=YES • 2=GIAL	V A R I A AT 2=NC	NCE FOR FOURTH	GRADE PUB	PILS
URES N DAT AURD TREAT ELNOI ELNOI ETMNI XVIRA	IE = UI/2 KNJALUU MENT I=(STATUS) R 1=2001 IC ITY I=	25/79) 578 5JNC+2=5 1=Y=5 25/79)	V A R I A AT 2=NC	\ N C ⊑ ★ ★ ±	r 14 44 47 175 4	• • •
URES N DAT AURD TREAT ELNOI ELNOI ETMNI XVIRA	IE = UI/2 KNJALUU MENT I=(STATUS) R 1=2001 IC ITY I=	25/79) 578 5JNC+2=5 1=Y=5 25/79)	V A R I A AT 2=NC	N C ⊑ ★ ★ 4	г м м ж (
AURU TREAT EUX S GENDZ ETHNI XVLAA	KNJALUU [MENT 1=0 STATUS 51 LR 1 =304 [CITY 1=5	5 78 CJNC+2=5 • 1=Y=5 • 2=GISL	AT 2=NC	N C ⊑ ★ ★ 4	к 41 47 т. (* * * *
EUY S GENDE Ethni Yvira	57ATUS = 1 22 2 = 304 - 1 C I TY 1.=	1=YES 2=GIEL	2=NC			
¥V1_***						:
	KNUNLUU	아트 이토트린트	(-2817 MATU * * * * * *	.S	ie s .	
•	~ ~ ~ ~ ~			•		
		JQ _ J	DF	SOU ARE	F	51 GN1 F 07 =
٩		315.752	5 L	1176.650	23.644	0,007
	1	1,332	I 1 3	5408,152 1,332 64,493	0-027	0 · 0 0 · 471 0 · 275
		214,145	2 1 1	420,230 214,145 537,097	4- 274 10-720	0 : 000 0- 042 0- 002
	1	101.395 36.385	12	71,793 36,395	15 932 Co 726	0,040
	·	660121 490811 280809	1 3 1	66,121 16,604 28,909	1 • 320 2• 331 0 • 575	0 - 255 0 - 203 0 - 451
			3	77.845	1.753	0-12- 0-109
JU 0 U 3 7		4,441	7	120,088 4,441 248,068	2-307 0:0=7 1,951	0,130,-
207 1007	,		. 2	125,325	2- 501	0,477 0,567
ຽງລາ		46n 531 46n 531	. ► 1	46,531 46,531	0, 22 n 0, 22 n	0,333
	3 3	901,035	23	353, 504	. 0	0 - 000
	د ۲۰۰۰	436.973	. 69	50,101		
	د 1	335,003	97	137,711		
	▲ J T D ↓ J U D ↓	AUTD KNU NL UU C 本 本 本 で G UU UU UU UU UU T UU T UU T UU T UU T UU T UU T UU S S S S S S S S S S S S S	AURD ANUNLOUE 77 3 UM DF 5 OUANES 705 C, e 3A 375 752 04 0 3 US2 1, 332 1 7 3 477 952 573 214, 145 5 37 0 C 77 1101 375 30 305 660121 490 H11 20807 293 534 315, 604 940 513 404 6135 12, C 31 UU7 250, 651 460 531 460 531 460 531 460 531	AURD ANUMLOUE 77 3000 ANUMLOUE 77 3000 ANUMLOUE 77 3000 ANUMES OF 7050, 034 ES OF 3053752 1 04030152 1 1,332 1 173.477 3 952.593 2 214.145 1 537.0797 1 1101.395 12 36.365 1 66.121 1 49.811 3 20.807 1 203.634 3 315.604 3 540.613 7 4.441 1 007 4.76.135 2 005 4.6.531 1 46.531 1 46.531 1 1005 4.6.531 1 46.531 1 1005 4.6.531 1 1005 4.6.531 1 1005 4.6.531 1 1005 4.6.531 1 1005 4.6.531 1 1005 4.6.531 1 3456.573 59	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	AURU ANJALUJE 77 3 304 0F 5 00 4RES DF 7050, 934 3 35, 752 1 395, 752 1 332 1 1, 332 1 1, 332 1 1, 332 1 1, 332 1 1, 332 1 1, 332 1 1, 332 1 1, 332 1 214, 145 1 36, 395 0, 776 66, 121 1 66, 121 1 66, 121 1 36, 395 1 36, 395 1 36, 395 1 28, 909 0, 575 2 3, 634 3 105, 201 2 4, 906 1 2, 135 2 49, 063 1 26, 907 0 0, 7 1 2, 031 2 4, 040 1 2, 037 1 2, 037 1 2, 031 2 4, 040 1 2, 037 1 2, 037 2 3, 537 5 7, 059 5 0, 101 5 7, 059 5 7, 059

ERIC Full Text Provided by ERIC

102 CASES WERE PERCESSED. 4 CASES (3,7 PCT) HERE MISSING.



1

· · ·	H.2 FOUR-WAY	ANALYSIS OF COV	ARTANCE FOR	FOURTH GRAD	E PUPILS
GRADE 4 ANDVAS PPL DA GRADE 4 ANDVAS. P FILT PDV4 (C ²)	NTA NEW COVAR DSTRCCAES TATION VATE =	IALLES MODEL 2 01/25/791		, , , , , , , , , , , , , , , , , , ,	
V177) 37 V177 XTDY V170 V170 V170 XTDY V170 X171 V171 V170	A THEAT AGAT GUY STATU GUY STATU A GUNDER 2= 7 ETHNICITY AATL VARIAGILI 419 TUTAL MIN	UMP 78 1=CINC+2=54T 5 3+ 1=Y55 2=* 0JY+ 2=GIRL 1=SPN 2=NGR 3 TY IN USE CF C UT_5 NONINSTR	SECAU 4=OT	' H	
	* * * * * * * *	* * * * * * * *		• • • •	
SOURCE OF VARIATION	•	50M 0= 5004= 35	df	9 C J 6 7 F	501GNIE E 05 5
4414 SFF ICTS 943004 XEDY 943005 943005	•	73360213 2229924 73519535 50504 1669025	6 1 1 3	1314,45° 222,924 7351,53° 5,504 55,342	13,042 0.009 2,705 0.105 97,321 0.0 9,967 9,777 9,671 0,573
COVAPIATES VOIFMATL STIMNIN VARO12		1632.410 129.233 506.339 1157.120	3 1 1 1	546,137 129,253 506,339 1157,120	5,527 0,001 1,546 0,215 5,144 0,016 14,040 0,000
2- XAY INTERACTIONS VAROUA XEDY VAROUA VAROOG VAROJA VAROOG XEDY VAROO7 VAROO6 VARCO7	-	972,672 3,586 29,774 147,460 3,520 590,250 95,132	12 1 1 3 1 3 3	82,723 31536 29,774 49,153 3,580 196,750 31,711	1.004 0,454 0.044 0,937 0.341 0,450 0,564 0,937 0,564 0,520 0,564 0,520 0,720 0,935 2,387 0,077 0,385 0,751
J- #1Y INTERACTIENS VAR004 XEDY VAR004 XEDY VAR004 VAR006 XEDY VARCO6	V 17 UUC V 17 UUC V 17 UU 7 V 17 UU 7	533,367 Uo655 1320294 253,703 B,175	7 1 2 2 2	76, 125 0, 556 65, 147 129, 351 4, 033	0, 22 0, 434 0, 009 0, 72 0, 903 0, 153 1, 569 0, 216 0, 050 0, 952
4-444 INTERACTIONS VARU04 XODY VARU04 VARU7	VAFJJÖ	219.012	1	219,012 213,011	2,645 0,103 2,645 0,107
TXPLAINED	;	11269,273	29	338, 595	4.715 C.COD
RESTOUAL		5274,641	64	82,416	\mathbf{x}_{i} , \mathbf{y}_{i} , \mathbf{y}_{i}
TOTAL		10543,914	ġ3	177.892	1

. 113

<u>۰</u>،

- «	H.3 FOUR-WAY	ANALYSIS OF COV	ARIANCE FOR	FOURTH GRADI	PUPILS	
		AULES NODEL 2			· .	•
	TON DATE = 0		_ (j#			
• • • • • • • • • • • • • • • • • • •	EDY STATUS GENDER 1=6 STRNICITY VANIABILIT TUTAL REAU NI NU	I.G 7E 1=C JNC 2=SAT J. 1=YI3 2=M J. 2=GIRL 1=SDN 2=NJR 3 Y IN USE JF D ING 77 ING 77 INS 77	ECAU 4EDTH IFF MATLS	<u> </u>	, 4 η 2 π δτ. 	• • • •
XNOCLE XUSECY	t hean hu jf	ADDES CODED		ez er et er⊨o	* * * *	* * *
SOURCE OF VARIATION		SUM JE SUUAFES	0F	MEAN SQUARE	F	516415 CF F
MAI 1 TFFECTS VARUO4 AIDY VARUU5 VARU05 VARU07		7800+483 305+942 7233+063 10+375 190+729	ĩ	305,742	30, 607 7-273 72, 044 0, 255 1, 511	C, COO 0,007 0, £13 0, £13
COV 15 1 ATES VOIFMATL VAROLE STI 4NIN KNODLES XUSEDY		1937,922 117,201 1372,694 434,772 200,831 51,993	5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	377,594 117,201 372,403 434,772 200,331 351,983	2,724 2,724 22,432 10,311 3,775 1,236	C, COO 0, 100 0, 000 0, COP 0, 0.33 0, 271
2-WAY INTIRACTICNS VAROUA XTDY VAROUA VADODS VAROUA VADODS VAROUA VADOT XIDY VADOOS XIDY VADOOS XIDY VADOOS VADOS VADOOS	· · ·	616,772 18,381 10,52 34,075 6320 476,775 77,733	12 1 1 3 1 3 3	0,320 159,502 119,331 10,502 11,355 0,320 159,502 25,714	1.619 0.25 0.25 0.70 0.70 0.75 0.616	0,110 0,511 0,617 0,947 0,731 0,015 0,607
3-#4Y INTEGAITIINS V47004 XEDY V42004 XEDY V42004 V47005 XEDY V47005	V A 7 JU 0 V A 7 JU 7 V A 7 JU 7 V A 7 JU 7 V A 7 JU 7	3 4 7 ° 7 5 0 0 7 6 4 7 2 0 1 ° 7 2 5 3 5 ° 3 5 0 3 9 ° 5 5 1	* 7 2 2 2	49,879 0,547 100,943 17,925 19,275	1, 1 + 1 0, 01 = 2, 39 = 0, 4 = 4 0, 4 = 4	0- 327 0- 702 0- 637 0- 657 0- 635
A-WAY INTERACTIONS VAROOA XEDY VAROOA XEDY	V10006	45,232	i	45,932 45,920	1,000	0, 701 0, 301
EXPLAINED	• • • • •	102393134	31	351,506	P. 35 P	c, coo,
RESIDUAL		2506,003	61	4,2,065		
TOTAL		13465.191	<u> 52</u> -	146, 3*1		
	•		ß	, ** , **	,	-

ERIC Author Provided by ERIC

APPENDIX I: DIFFERENCE SCORE ANALYSIS

12-1

1.1 WORD KNOWLEDGE

I.2 TOTAL READING

		·		DIFFERNCE SCOR	ANALYSIS .	· · · · · · · · · · · · · · · · · · ·
57/10 E 6 10/2 E 10/2 E 10/2 E 10/2 E 10/2 E 10/2 E 10/2 E		LIPAPPEREN Ston Date :				· <
				V A Z I A '		*****
•	· · · · · · · · · · · · · · · · · · ·	THEAT HE	17 1=0 3:10+2	- 		
	x" 1 VA=) ;4	e sender a	L=Y=: 1=0_Y • L=GI	41	17-4	
1 / / · · · · · · · · · · · · · · · · ·	10074V2 10074V2		131 0F 80Y	152 3=240 4= MATERIALS		
		* * * * * * *	•••			- -
		•	ل الرو E + درو ال	- 	501477	e
TTURCE OF VE		•	51050J	j.	- 5.001	1-3-1 2-143
4111 TEFECT3 V12304	1			•	- J, C - 4 33, 247	1-257 01261
X = : Y V A = 0 7 7			30,;0 31,:0	3 1/	30,70ª 111,0°0	0-475 0-157
16-007			2.52 L		5,1)2	1-523 0-221
CIV 1-11717			9 tao 1 0	i X	25-102	1-523 0+221
2- 4 14 147.394	CTIONS	-	1300,05		103,704 N.3,775	1,724 0,075 1,742 0,153 0,354 0,554
V12004	VARUC5 V42007		22012)5 I (1) J	22,125	2, 472 2
XLTY	V17014 V17014		5 ÷ د 7 5 1. به د ۳ 5 2 ت	J 3	37, 442 109, 327	0, 1, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0, 0,
	VA2007		3-0,27		125,755	2,055 0-055
3- WAY 147534 VAROO4	CTIONS XTOY	マハマビレコ	31304	1 1	130,434 33,351 1370534	2-202 0-113
A73004	XEDY	ジュマンリア シュニンフ	27 E o C E 4 0 o A J	3 2	20.307	0,725 0.723
· V4=004 KEDY	VA= 006	y 3 0 J J 7	32132			C5 51 3 \ C 5 5 2 5
A-NAY INTERA VEROCA	x 71713	127005	32503 32503		32,052	0.513 0.574
• • • • • • •	14=0)-				1:)5,617	1-500 3-241
TXPLAINED			2357,0	·	£ 2, 44 ¢	
77512UAL		· · · · · · · · · · · · · · · · · · ·	4371+3		7-1-70	
7074L		•	7223,4	· • • • •	,	
102 015		a:pc73520,	1	. .		·
4 C2 3	== (3,	a 7(7) wiki	MISSINGA		• /)	с.,

Ν.

Full least Provided by ERIC

đ

o

121 :

57407 \$ PUPIL 047	A 7= 1° 1°N DATE = J	1/23/79)	•		•	
))* T+E+T+FENT (ニンドー>T+FU=))+ CE+ND2++ 2 ==	1=CONC+2=5A1 5 3 6 1=¥35 2: 5 34, 2=619L	=NC /		• • • • •	r r • •
WITH STI WITH STI X'77 X'77 VOIT V27	1911 Y. TUTAL AIRU 2017 - MEAN AU UM 507 - Averaue US 774 - VNiruru (1	1=SPN 2=NGP JTES NONINSTR - 20125 00020 JE OF 204 430 TY IN USE 77 JUS 77	R TIME D TERIALS	,	- 6	
	• • • • • • • • • • • • • • • • • • •	4 3 8 8 8 4 4 	****	₹ ₽ ₹ ₩ ₩	τ ε τ ε τ ^ε ι τ	* * +
SJUPCE OF VARIATION	•	SUN 19 SUU4423	2F *	NEAN SULARE	E	SIGNIE FE E
411 11 7FF 2073 V2=004 V2=004 V2=007 V2=007		372,33) 37,753 73,755 50,423 207,630	5 1 1 3	54,723 39,743 53,754 60,499 59,217	1,530 0,54 1,501 1,43 1,645	2 · 1 · 1 C · 33 · O · 1 · 3 C · 23 · O · 1 · 3
COVARIATES STIMNIN KNROLES KJSEDY VDIEMATL VAROLZ		1400, C53 4540773 2002932 51.4323 117,203 474,374	5 1 1 1 1 1 1	272,011 454,773 200,322 51,234 117,203 674,376	6, 342 10, 411 4, 77 1, 234 2, 794 1, 032	0,000 0,002 0,033 0,271 0,100 0,100
2-414 INTERACTICNS V12004 XEDY V12004 V12007 V12004 V12007 XEDY V12007 XEDY V12007 V12005 V12007		9 L 4 e 9 7 3 L 1 7 J 3 1 L 0 e E J 3 3 4 e 0 7 4 C : 3 2 J 4 7 2 7 7 4 1	12 1 3 1 3 3	50003 185391 105503 115355 05320 1595594 25,714	1 - 61 G 0 - 63 7 0 - 25 0 0 - 7 7 0 3 - 7 7 6 0 - 61 6	0,110 0,511 0,511 0,547 0,731 0,731 0,607
3-#14 14758467178 V1-004 XF04 V1-004 X'04 V1-004 V1-004 KEDY V4-00-	V 1 = JJ = V 1 = JJ = V 1 = JJ = V 1 = JJ 7	347,747 Upia7 2010722 35,630 34,745	7 1 2 2 2	49, 679 0, 617 100, 961 17, 725 19, 273	1,1P1 C-015 2,350 C-426 0,450	0 · 127 6 · 702 0 · 1033 0 - 155 0 · 137
4-#14 1417594671 CN3 V4 2004 XEDY V4 2004 XEDY	VAFUUt	4 50 323 4 30 427	i 1	45,825 45,827	1,0*0 1- C°C	৩ - 301 0 - 301
EXPLAINED		3057,5±4	31	1981 676	2,34=	0,002
FISTOUAL		25550735	21	42,065		
TOTAL		5624, 453	92	÷1,141		

122

ERIC FullTaxt Provided by ERIC